

ADM 12 Video Display Terminal Users Reference Manual

LEAR SIEGLER, INC. DATA PRODUCTS DIVISION 901 EAST BALL ROAD, ANAHEIM, CALIFORNIA 92805

ADM 12 Video Display Terminal



USERS REFERENCE MANUAL

DP2980984F

©1984 LEAR SIEGLER, INC.

SEPTEMBER 1984

LEAR SIEGLER, INC.

TABLE OF CONTENTS

SECTIO	N	PA	GE
Ι	GEN	NERAL DESCRIPTION	. 1-1
	1.1	INTRODUCTION	1-1
	1.2	OPERATIONAL DESCRIPTION1.2.1Keyboard1.2.2Primary (Modem) and Auxiliary Ports1.2.3Set-Up Mode1.2.4Programmable Function and Edit Keys1.2.5Power Requirements	.1-2 .1-2 .1-2 .1-2
	1.3	SPECIFICATIONS	1-2
	1.4	ADM 12 WORD STRUCTURE	1-2
	1.5	ANSWERBACK MEMORY	1-3
	1.6	OPTIONS1.6.120mA Current Loop1.6.2RS-4221.6.3International Keycap/Character Generator Sets1.6.4Amber Phosphor Display1.6.514-Inch Display	.1-3 .1-3 .1-3 .1-3
II	INS	TALLATION	. 2-1
	2.1	INITIAL PREPARATION2.1.1Line Voltage Selection2.1.2Set-Up Mode Preparation2.1.3Keyboard Plug-In2.1.4Monitor Keyboard Adjustment	.2-1 .2-1 .2-1
	2.2	INTERFACE INFORMATION2.2.1RS-232C Interface2.2.220mA Current Loop Interface2.2.3Auxiliary Port Interface	.2-2 .2-2

TABLE OF CONTENTS (cont'd)

PAGE

2.3	INSTALLING THE ADM 12 2-4
2.4	POWER TURN-ON
2.5	SCREEN SAVER 2-5
2.6	POWER TURN-OFF 2-5
2.7	CARE OF THE ADM 12 2-5 2.7.1 Cleaning 2-5 2.7.2 Inspection 2-5

III	OPI	ERATION
	3.1	KEYSTROKE CONVENTIONS
	3.2	SET-UP MODE3-13.2.1Selecting Functions3.2.2Saving Set-Up Functions3.2.3Down-Line Load Set-Up
	3.3	OPERATIONAL MODES3-63.3.1On-Line Or Local Mode3-63.3.2Conversation or Block Mode3-63.3.3Duplex Operation3-63.3.4Program Mode3-6
	3.4	KEYBOARD OPERATION3-63.4.1Keystroke Entry3-103.4.2Alphanumeric and Punctuation Keys3-103.4.3Numeric Keypad3-103.4.4Modifier Keys3-103.4.5Cursor Control Keys3-103.4.6Edit Keys3-113.4.7Transmission Keys3-123.4.8Programmable Function Keys and Edit Keys3-133.4.9Special Operation Keys3-15
	3.5	DISPLAY CHARACTER FORMAT3-163.5.1International Fonts3-16
	3.6	COMMAND CODE SET 3-16 3.6.1 Control Codes 3-16 3.6.2 Escape Sequences 3-16

TABLE OF CONTENTS (cont'd)

SECTION

PAGE

3.7	DISPL 3.7.1 3.7.2 3.7.3 3.7.4 3.7.5	AY FORMAT. Vertical Scrolling Horizontal Scrolling Selectable Right Margin Screen Video Attributes Personality Character Selections	3-27 3-27 3-27 3-33
3.8	DATA 3.8.1 3.8.2 3.8.3	TRANSMISSION Conversation Mode Block Mode Handshaking Protocol	3-34 3-34
3.9	CURS(3.9.1 3.9.2 3.9.3	OR CONTROL Relative Cursor Positioning Absolute Cursor Positioning/Reading Tab Control	3-34 3-34
3.10	GRAP	HICS	3-35
3.11	SET-U	P MODE OPERATIONS	3-36
3.12	PRINT	OPERATIONS	3-36
3.13	RESE7 3.13.1 3.13.2 3.13.3 3.13.4	COPERATIONSPower-On ResetReset Terminal CommandSelf-TestStatus Information	3-36 3-36 3-36

TABLE OF CONTENTS (cont'd)

LIST OF FIGURES

FIGURE		PAGE
1-1	ADM 12 Video Display Terminal	1-0
2-1	ADM 12 Dimensions	2-2
2-2	ADM 12 Controls and Connectors	2-3
2-3	Typical ADM 12 Applications	2-4
2-4	Modem Connector and Auxillary Port	2-6
2-5	Optional Current Loop and RS-422 Interface Logic	2-7
3-1	Sample Set-Up Mode Functions	3-3
3-2	ADM 12 Set-Up Mode Display And Associated Keys	3-4
3-3	Programmable Edit Keys	3-5
3-4	ADM 12 Standard Keyboard Operation Characteristics	3-7
3-5	Factory Set Default Condition of Programmable Keys	3-14
3-6	ADM 12 Display Memory Configurations	3-28
3-7	Set-Up Mode Flow Chart	3-38

LIST OF TABLES

TABLE	PAG	E
1-1	ADM 12 Specifications 1-	4
3-1	Keystroke Conventions 3-	1
3-2	Control Codes Utilized by the ADM 12 3-1	7
3-3	ADM 12 Escape Sequences 3-1	9
3-4	Embedded Attributes Operation Codes 3-33	9

APPENDIX

TABLE		PAGE
A-1	ADM 12 Set-Up Mode Functions	A -1
A-2	Binary Set-Up Functions	
A-3	ADM 12 Escape Sequences	
A-4	Reference Chart of Absolute Cursor Positions	
A -5	ASCII Control Code Chart	A -10
A- 6	Displayable USASCII Character Set and Control Codes	A-11
A -7	128 Character ASCII Format with Hex Codes	A-12
A-8	Business Graphics Character Set	A-13
A- 9	International Keyboard Layouts	A-14
A-1 0	Model Configuration Identification	A-16

PREFACE

The ADM 12 Video Display Terminal is a fully featured general purpose ergonomic, conversational and block mode terminal, ideally suited for a wide range of applications.

This Users Reference Manual explains how to use the ADM 12, and provides information on its installation and care. To help you effectively use the great variety of ADM 12 operating features, this manual emphasizes the various features and control codes.

Lear Siegler Inc./Data Products Division constantly strives to maintain the accuracy and useability of our technical publications. We welcome your comments and suggestions, and have provided a pre-paid, pre-addressed post card at the back of this manual to note any changes you feel will improve this publication.

WARNING

This equipment generates, uses, and can radiate radio frequency energy and if not installed and used in accordance with the instruction manual, may cause interference to radio communications. It has been tested and found to comply with the limits for Class A computing devices pursuant to Subpart J of Part 15 of FCC Rules, which are designed to provide reasonable protection against such interference when operated in a commercial environment. Operation of this equipment in a residential area is likely to cause interference in which case the user at his own expense will be required to take whatever measures may be required to correct the interference. Only shielded cables with the shield terminated to the metal hood of the connector can be used.



Figure 1-1. ADM 12 Video Display Terminal

SECTION I GENERAL DESCRIPTION

1.1 INTRODUCTION

The Lear Siegler ADM 12 Video Display Terminal, shown in **Figure 1-1**, is a smart, editing terminal designed for user convenience. It can be used in either the Conversational or Block mode of operation. The non-volatile memory retains: the desired set-up configuration, the information in the 16 programmable Function Keys (shiftable to 32), and the 6 Programmable Edit Keys (shiftable to 12). All the terminal set-up functions are soft-setable from the keyboard. Function keys and Edit keys can be programmed by the host or from the keyboard. For added user convenience, the ADM 12 offers three display memory configurations: 24 lines by 80 characters, 48 lines by 80 characters, and 24 lines by 158 characters. The ADM 12 is also ergonomically designed for maximum user comfort. Its features include: a tilt and swivel monitor with positive detents; a low-profile, user-friendly, DIN standard keyboard; a six foot long coiled keyboard cord. Designed for speed and versatility, the ADM 12 is ideal for most data terminal applications.

1.2 OPERATIONAL DESCRIPTION

The ADM 12 is used to enter, display, print, and send asynchronous data. It is also used to receive, print, and display data from an auxiliary device such as a host computer. Its most frequently used application is one in which an operator interacts in a two-way communication with a host computer. However, in some applications the data transfer may be unidirectional either from the ADM 12 to the host or from the host to the ADM 12. The ADM 12 is a complete stand alone data terminal consisting of a 12-inch (30.5 cm) or optional 14-inch (35.5 cm) CRT Monitor Unit and a detached, low-profile keyboard unit. Each unit is contained in a strudy enclosure which also houses the required electronic circuits. Some of the features provided in the ADM 12 are:

- Full 128 ASCII character set
- Three types of display memory configurations:
 - 24 lines by 80 characters used for general purpose terminal applications
 - 48 lines by 80 characters for applications such as word processing
 - 24 lines by 158 characters for applications such as financial reporting and spreadsheet applications.
- DIN-standard keyboard, with:
 - Numeric keypad with 14 keys
 - Five separate cursor control keys arranged in a cross pattern
 - Sixteen programmable function keys (shiftable to thirty-two)
 - Six programmable edit keys (shiftable to twelve)
 - Cap lock key
 - Status key
 - Hold Screen key
 - 3-key rollover and 32 keystroke buffer
 - User friendly layout and rib dividers to reduce operator error
 - Tilt mechanism
- On-Line or local modes
- Conversational or block modes
- Program mode

Display

- Split screen with scrolling region
- Vertical and horizontal scrolling
- Selectable right margin
- Full screen reverse video
- Five non-embedded attributes:
 - Blank
 - Blink
 - Reverse
 - Underline
 - Reduced intensity
- In embedded attribute mode select four of five embedded attributes:
 - Blank or underline
 - Blink
 - Reverse video
 - Plus non-embedded reduced intensity
- 7 x 10 character matrix
- Protected fields
- Business graphics
- Absolute cursor positioning
- Variable 7 or 8-bit data word structure
- Seven communication rates in full-duplex and half-duplex send/receive modes
- RS-232C modem interface and RS-232C auxiliary interface port
- Four print modes:
 - Page Print
 - Line Print
 - Display and Print from host
 - Transparent Print from host
- Non-volatile memory for:
 - Set-Up modes
 - Programmable Function keys
 - Programmable Edit keys
 - Answerback Memory

1.2.1 Keyboard

The keyboard, which meets DIN standards for operator comfort, is attached to the rear of the monitor with a 6-foot coiled cord to provide complete keyboard mobility.

1.2.2 Primary (Modem) and Auxiliary Ports

The **ADM 12** comes standard with two ports: the RS-232C Modem Port and Auxiliary Port which are located at the back of the terminal. The Modem (I/O) Port is the link, or interface, that handles the flow of data in both directions between the **ADM 12** and the host computer.

The Auxiliary Port permits the use of an RO (Receive Only) printer when hard copy of data is needed.

1.2.3 Set-Up Mode

The **ADM 12** features a non-volatile Set-Up Mode which allows the terminal parameters to be selected through commands from the keyboard or host. These parameters, including baud rates, word structure, cursor type, communications control, and operating modes, can then be "saved" in memory, and retained when power is shut off.

1.2.4 Programmable Function and Edit Keys

The **ADM 12** has sixteen programmable Function keys (shiftable to thirty-two) and six programmable Edit keys (shiftable to twelve). The information user-programmed into these keys can be saved in non-volatile memory.

1.2.5 Power Requirements

The **ADM 12** can accept 115 VAC or 230 VAC +/- 10%, 50 or 60 Hz. The voltage is factory set, and must be specified at the time of order.

1.3 SPECIFICATIONS

The ADM 12 specifications are listed in Table 1-1, page 1-4.

1.4 ADM 12 WORD STRUCTURE

The **ADM 12** transmits serial asynchronous data in a 10 bit format in the sequence: one start bit, seven or eight data bits, one or no parity bit, and one stop bit. Parity selection (odd, even, or no parity) is permitted only with a seven bit data word. If an eight bit data word is selected, then bit eight can be designated to be either 1 or 0. Parity selection and the value of bit eight are defined via Set-Up Mode. The received data shall be formatted the same as the transmit data. The word structure, baud rates, and other communications characteristics are established via Set-Up Mode selections.

1.5 ANSWERBACK MEMORY

The ADM 12 provides a 32 byte, userprogrammable, Answerback message stored in non-volatile memory. The message is transmitted in response to an ENQ code from the host computer or the command CTRLE from the keyboard.

1.6 OPTIONS

The standard **ADM 12** can be further equipped with several options.

1.6.1 20mA Current Loop

The **ADM 12** with the optional 20mA current loop interface can operate at a maximum baud rate of 9600 from the Modem Port. This option is plug mounted to permit field installation.

1.6.2 RS-422

The logic board design permits RS-422 to be added as a field or factory installed option. This option is as defined in E.I.A. Specifications RS-422 (**Figure 2-5, page 2-**7).

1.6.3 International Keycap/ Character Generator Sets

Several character sets are optionally available with the **ADM 12**, as well as the associated keycap sets. These include UKASCII, French, German, Swedish, Finnish, Danish, Norwegian and Spanish.

1.6.4 Amber Phosphor Display

While the standard **ADM 12** is equipped with a P31 green phosphor display, an amber display is optionally available in place of the green CRT at no additional cost when specified at the time of order.

1.6.5 14-Inch Display

The **ADM 12** is available with an optional 14inch (35.6 cm) CRT screen with green or amber phosphor with non-glare surface.

LEAR SIEGLER, INC.

Table 1-1. ADM 12 Specifications

SPECIFICATION	DESCRIPTION	
DISPLAY	· · · · · · · · · · · · · · · · · · ·	
CRT Screen	12-inch (30.5cm) diagonal; green phosphor with non- glare surface. (Amber optional.) Optional 14-inch (35.6cm) display with green or amber phosphor with non-glare screen.	
Display Area	12-inch screen: 8.25" W (21cm) x 5.75" H (14.6cm) 14-inch screen: 9.05" W (22.98cm) x 6.55" H (16.64cm) ± .05 (.13cm).	
CRT Console	Tilt and Swivel mechanism with positive detents. 360° swivel capability; tilt 5° forward to 15° back of vertical. Removable base.	
Horizontal Refresh Rate	17.7 kHz	
Vertical Refresh Rate	50 Hz or 60 Hz, depending on line frequency, factory set at time of order.	
Display Formats	 24 lines of data 80 characters wide can be displayed plus 25th line for status. Three display memory configurations are available through set-up as follows: 1. Two independent displays each 24 lines by 80 characters. 2. One page, 48 lines by 80 characters. 3. One page, 24 lines by 158 characters. 	
Display Page Density	1920 characters (2,000 with status line)	
Character Font:		
Character Field	8 x 11 dot matrix.	
Character Matrix	7 x 10 dot matrix with descenders.	
Cursor	8 x 11 steady or blinking block or underline; can be turned off or on.	
Character Sets	128 displayable characters including control codes. Optional international character sets available.	
Business Graphics	Block Graphics, Wide Point Graphics, and Line Drawing characters.	
Visual Attributes:		
Non-Embedded	Five non-embedded attributes: blank, blink, reverse video, underline and reduced intensity.	
Embedded	Four of five embedded attributes; blank or underline depending on set-up, blink, reverse video and reduced intensity.	

SPECIFICATIONS	DESCRIPTION	
Visual Attributes (cont'd)		
Protect	Write Protect on a character by character basis.	
Background Attribute	Full screen normal or reverse video.	
Formatting Aids:		
Program Mode	For displaying received control characters.	
Page Edit Mode	Insert/delete characters in a line causes automatic wrap around of lines following.	
Write Protect Mode	All characters on screen are Write Protected and displayed in the user selected "protected attribute".	
Protect Mode	Protected fields are protected from overwrite.	
Insert Mode	Characters are inserted at the cursor position as typed. Existing data is moved to the right.	
Scrolling Region Mode	Split screen scrolling limits may be set.	
Cursor Movement	Relative and absolute addressing, and read cursor cap- abilities.	
KEYBOARD FUNCTIONS		
Keyboard	Detached, low-profile, DIN standard with six-foot coiled cord. Sculptured keys, selectric-type layout. Calculator format numeric keypad. Two tone keycaps, by function; rib dividers to separate three major groupings of keys. 11° keyboard tilt mechanism. Keys are auto-repeating (15 chars. per second). 32 character keystroke buffer.	
Keyboard Layout	QWERTY - Typewriter Pairing. Optional QWERTY - teletype pairing, QWERTZ and AZERTY.	
Numeric Keypad	14 keys, 0 through 9, Enter, comma, period, and minus. 0 and Enter keys are double-size.	
Cursor Control	Up, down, left, right, home arranged in a "cross" pattern. Return, Line Feed, New Line, Backspace, Tab and Back Tab.	
Function Keys	Sixteen programmable function keys, shiftable to thirty- two functions each transmitting up to a 4-character se- quence with either local or host destination code. Non- volatile memory storage.	

Table 1-1. ADM 12 Specifications (continued)

Table 1-1	ADM 1	2 Specifications	(continued)
-----------	-------	------------------	-------------

SPECIFICATIONS	DESCRIPTION
KEYBOARD FUNCTIONS (cont'd)	
Edit Keys	Six programmable edit keys, shiftable to twelve. Non-volatile memory storage.
Function Command Keys	Escape, Set-Up/Status, Clear/Delete, Aux Port, Hold Screen, Break, Cap Lock, Shift, Control and more.
Non-Volatile Set-Up Mode	No dip switches - Terminal functions are selectable from the keyboard, or remotely from the host. Features se- lectable in Set-Up Mode include: Key-Click, Communi- cations Characteristics, Replacement Characters, Com- munications Configuration, and more.
Operator Convenience	Selectable key-click for audible feedback and 3-key roll- over to eliminate "missed" characters. Full screen re- verse video selectable cursor.
EDITING	
Edit Operations	Editing can be from the keyboard or host. Editing function includes clear screen, erase to end of line/page, Insert character/line, page edit, and more.
TRANSMISSION MODES	
Conversation Mode	Interactive (character-by-character transmission, Full or Half Duplex operation).
Block Mode	Data entered on terminal is processed locally and trans- mission to the host is accomplished by various sends available, such as: Send Line, Send Page, Send Message.
Control Characters	Control characters may be transmitted to computer and/ or entered on screen.
Interfaces	RS-232C or optional 20mA current loop or optional RS-422.
COMMUNICATIONS	
Primary Port	RS-232C, serial asynchronous ASCII communications.
Auxiliary Port	RS-232C with page print, line print, display and print and transparent print.
Data Rates	300, 600, 1200, 2400, 4800, 9600, and 19,200 baud (9600 max. for 20mA current loop)

LEAR SIEGLER, INC.

Table 1-1. AD	M 12 Specification	s (continued)
---------------	--------------------	---------------

SPECIFICATION	DESCRIPTION
COMMUNICATIONS (cont'd)	
Word Structure	1 start bit, 7 data bits, 1 parity bit (odd, even, mark space or none - selectable), and 1 stop bit.
Parity	Even, odd, or no parity.
Stop Bits	1 stop bit.
Format	ASCII serial asynchronous communications.
Busy Indication	Primary (Modem) Port: Sends X-OFF (DC3), X-ON (DC1) or DTR to host on busy/ready condition of main port or auxiliary port.
	Auxiliary Port: Senses busy level on Pin 20 (DTR) of the Auxiliary interface.
Input Buffer	1024 bytes.
Handshaking	X-ON/X-OFF (user-selectable characters) or, DTR (Pin 20) may be used to indicate the terminals Busy/Ready status or no handshaking.
GENERAL	
Operating Environment:	
Temperature	5° C to 40° C (41° F to 104° F).
Humidity	10% to 85% without condensation.
Power Requirements (Factory set)	115 VAC ±10%, 50/60 Hz 230 VAC ±10%, 50/60 Hz 50 watts

SECTION II

2.1 INITIAL PREPARATION

The space-saving design of the **ADM 12** has proven to be beneficial for almost every terminal site. Its small size is ideally suited to those work environments where physical space is limited.

The physical dimensions of the ADM 12 are shown in Figure 2-1, page 2-2. The basic requirements for installation are as follows:

- Table or desk mounting
- Standard three-pronged 115-volt (230-volt) A.C. power outlet.
- Cable connection to the computer, data set or modem; serial printer, or other auxiliary device. If connection to a remote computer is desired, a modem or data set is usually required.



Allow the ADM 12 to adjust to severe changes in the environment before applying power. This will reduce the possibility of internal condensation, which could impair its operation.

2.1.1 Line Voltage Selection

The ADM 12 is shipped ready to operate at either 115-volt or 230-volt AC as specified on the purchase order. Contact your authorized Lear Siegler maintenance personnel if the operating voltage needs to be changed.

2.1.2 Set-Up Mode Preparation

The operating configurations of the ADM 12 are controlled by firmware "switches" and have been pre-set at factory to the default conditions as indicated by "*" in Table A-1, page A-1 of the Appendix. The operating configurations are displayed on the status line when the Set-Up Mode is entered.

There are two levels of Set-Up Mode: operator convenience and supervisor set-up.

Operator Convenience set-up functions are characteristics that relate to operator comfort or preference. The Supervisor Set-Up functions include all of the feature selections available to the operator, as well as those functions useful to the person programming the system (see **paragraph 3.2**, **page 3-1**).

Normally, the Supervisor set-up functions are selected when the terminal is installed, or a new peripheral device is attached. **Table A-1**, **page A- 1**, describes the Set-Up functions which can be selected for both modes.

The Set-Up functions can also be downloaded from a host computer, as described in **Table A-2**, page A-3 of the Appendix.

Parameters of the host computer must be determined before the terminal can be used. If changes need to be made to the Set-Up mode functions, refer to Section 3.1, page 3-1.

2.1.3 Keyboard Plug-In

Before applying power to the terminal, plug the keyboard coiled cord into the connector on the back of the monitor. Refer to **Figure 2-2**, **page 2-3** for connector location.

2.1.4 Monitor, Keyboard Adjustment

The ADM 12 monitor can be tilted and swiveled to position it in the optimum viewing angle for the operator. Adjustments can be made by pushing down from the top of the monitor. "Clicks" heard, as the monitor is adjusted are from the positive detent mechanism, which holds the monitor securely in the desired position.

The keyboard may be tilted for maximum operator comfort. Simply turn down the two feet located under the rear of the keyboard housing, if desired.

2.2 INTERFACE INFORMATION

The ADM 12 may be cabled directly to a local computer, serial printer (or other auxiliary device), or it may be connected via telephone data lines to a remotely located computer. Remote computer connections require the use of a modem or data set. Figure 2-3, page 2-4 shows a typical ADM 12 application. Figure 2-2, page 2-3 shows the rear panel locations of the interface connectors. Figures 2-4 and 2-5, pages 2-6 and 2-7 show the logic associated with these interfaces.

2.2.1 RS-232C Interface

The Primary (Modem) Port RS-232C Interface provides the signals and levels associated with RS-232C, allowing direct connection to a computer or modem. The maximum permissable cable length is 50 feet for RS-232C applications.

2.2.2 20mA Current Loop Interface (Option)

The ADM 12 can as an option be configured for 20mA Current Loop operation using the Primary (Modem) Port. The current loop interface signal levels allow cable lengths of up to 1000 feet. The maximum baud rate for current loop operation is 9600.



Figure 2-1. ADM 12 Dimensions



Figure 2-2. ADM 12 Controls and Connectors

2.2.3 Auxiliary Port Interface

The Auxiliary Port is used for connecting an RO (Receive Only) serial printer or other RO device to the terminal using RS-232C signal levels (refer to **Figure 2-4**, **page 2-6**).

Note

The peripheral device that is attached to the Auxiliary port must operate at the same baud rate as the communication line.

2.3 INSTALLING THE ADM 12

- 1. Check the ON/OFF switch on front of unit. Refer to Figure 2-2, page 2-3 to ensure that it is set to OFF.
- 2. Connect the data interface cable(s) to the terminal using the appropriate interface information (**paragraph 2.2, page 2-**2).
- 3. Plug the power cord into a grounded AC outlet of the proper voltage.

4. Refer to Section III for operation and set-up configuration functions that might be desired such as baud rate selection.

2.4 POWER TURN-ON

- Set the ON/OFF switch on the front of the unit to ON (refer to Figure 2-2, page 2-3). A power on "beep" should occur within two seconds.
- 2. Wait approximately 20 seconds for the unit to warm up. The cursor should appear in the HOME position.
- 3. The Self-Test message " OK" should appear in the upper left corner and the terminal will sound the audible alarm.
- 4. If the Self-Test message and the cursor do not appear, turn off the **ADM 12** wait 15 seconds and apply power again. If the message and cursor still do not appear, check the CONTRAST control, then contact your authorized Lear Siegler service representative.



Figure 2-3. Typical ADM 12 Applications

5. Adjust CONTRAST control (Figure 2-2, page 2-3) for desired display brightness.



To avoid damage to the CRT screen when the terminal is to remain ON but unchanged for extended periods of time, reduce the screen brightness using the CONTRAST control on front of unit.

2.5 SCREEN SAVER

To prevent damage to the CRT screen, a Screen Saver feature is enabled as the factory default condition. If the screen display is not changed for approximately 15 minutes, the display disappears from the screen and is retained in memory. Depressing any key (or data received) will return the display to the screen. Screen Saver can be disabled in Set-Up Mode (see Appendix A, page A-1).

2.6 POWER TURN-OFF

If the unit is displaying information which must be retained, be sure to transmit this information to the host immediately. This is necessary, as the display is cleared when power is removed from the unit. An alternative is to send the information to a printer prior to turning the unit OFF.

2.7 CARE OF THE ADM 12

2.7.1 Cleaning

At periodic intervals, clean the exterior housing and lightly dust the unit using a soft brush or damp lint-free cloth or paper towels. Do not use petroleum base cleaners, such as lighter fluid, as this could be harmful to the housing. Remove smudges from the CRT exterior screen and housing with conventional spray cleaners or alcohol.

2.7.2 Inspection

The characters on the CRT screen should remain sharp, clear, and intense for the life of the terminal. If this should vary, your authorized Lear Siegler service representative can quickly make any required mechanical and electrical adjustments.



Figure 2-4. Modem Connector and Auxiliary Port

LEAR SIEGLER, INC.



Figure 2-5. Optional Current Loop and RS-422 Interface Logic

SECTION III OPERATION

3.1 KEYSTROKE CONVENTIONS

Commands entered on the **ADM 12** keyboard may require simultaneous depression of two or more keys, or sequential depression of two or more keys (one after another). The appropriate convention is indicated in this manual by how the command is shown graphically. **Table 3.1, below**, illustrates the two conventions.

3.2 SET-UP MODE

The ADM 12 has two set-up modes -OPERATOR CONVENIENCE SET-UP and SUPERVISOR SET-UP. The Operator Convenience Set-Up functions are characteristics that relate to operator comfort or preference. Supervisor Set-Up functions include all the features useful to the person programming the system. Alternately, all of the set-up functions can be downloaded from a host computer on a byte per byte basis. The OPERATOR CONVENIENCE set-up functions are accessed by entering [SHIFT] SETUP]. The SUPERVISOR set-up functions are accessed by entering SHIFT CTRL SETUP from the keyboard. Download of functions from the host computer is accomplished by an ESCAPE sequence as described in Sections 3.2.3, page 3-2.

Table A-1, page A-1 in Appendix describes the Set-Up mode functions that are configurable for the ADM 12. Figure 3-1, page 3-3, shows sample Set-Up mode functions.

3.2.1 Selecting Functions

When Set-Up Mode is entered, the status line (line 25) will disappear, and the first set-up mode function will be displayed in its place, (see **Figure 3-1, page 3-3**). The cursor will surround one of the available selections (e.g., N or Y); this is the current value associated with the function. To change the value associated with a particular function, press or \leftarrow until the cursor is positioned over the desired value. Press \square to display the next available function. For example, pressing \downarrow

Table 3-	1. Keystrok	e Conventions
----------	-------------	---------------

Keystrokes	Description
ESC A 2	Space between each key. These are independent keystrokes, typed one after the other.
SHIFT CTRL SET UP	No space between keys. These are simultaneous keystrokes. All three keys, SHIFT , CTRL and SET UP , should be depressed at the same time.

when CLICK is displayed will cause the STATUS function to be displayed if in the operator accessible mode. Pressing \uparrow will cause the previous function to be displayed. Pressing the <u>HOME</u> key causes the first function (CLICK) to be displayed.

Pressing the STATUS key will cause the terminal to exit Set-Up and return to the previously established mode. Before set-up is exited, the operator will be given the opportunity to SAVE the newly changed set-up features by the message SAVE?(Y/N). The operator must enter Y to save. The status line will appear at the bottom of the display screen after a "wait" period of approximately 6 to 10 seconds. The operator can go from operator to supervisor accessible functions while in Set-Up mode via the SHIFT|CTRL|SETUP| command mentioned above.

Pressing the STATUS key exits Supervisor Set-Up mode and causes any changes that were made to take effect and the status line, if enabled, will be displayed. There is no save message when exiting Supervisor Set-Up mode. The ADM 12 receives but does not process all received data while in Set-Up mode. When handshaking is enabled the ADM 12 will send an X-OFF or drop Data Terminal Ready if the host overruns the input buffer during Set-Up mode.

Note that if the terminal is currently in the Inhibit Personality Changes Mode (set-up byte 1, bit 7) the Supervisor Set-Up mode from keyboard is disabled and the current functions can only be changed from the host computer. Exiting from the set-up mode will result in the video and keyboard returning to previous conditions.

Note

Changing communications characteristics when in Set-Up Mode may affect or prevent further data transfer with the host computer or auxiliary device.

3.2.2 Saving Supervisor Set-Up Functions

When the various functions are selected, they take effect upon exiting Supervisor Set-Up Mode. The set-up is stored in temporary "working" memory and will be lost if the terminal is powered-down. To cause the set-up functions to be saved in non-volatile memory, enter the <u>CTRLIS</u> (press <u>CTRL</u> and <u>S</u> simultaneouly) command when in Supervisor Set-Up Mode.

There are two other commands that can also be performed in Supervisor Set-Up Mode: CTRLD and CTRLR. CTRL D causes the ADM 12 to be reset to the default function values. CTRLR causes the terminal to restore the function values that are currently in non-volatile memory (the current function values might not have been saved). A CTRL S must follow a CTRL D if the desired settings are to be saved in nonvolatile memory. A save, default, or restore operation will not be performed if the corresponding key sequence ([CTRL]S], [CTRL]D, or [CTRL]R]) is entered while the terminal is in any mode other than Supervisor Set-Up. The ADM 12 will always power-up to the last saved set-up conditions. See figure 3-7 on page 3-38 for more Supervisor Set-Up commands.

3.2.3 Down-Line Load Set-Up

The Set-Up mode features can be selected by the host computer by a command sequence on a byte per byte basis. This is accomplished via the escape sequence ESC { AA HH where AA = Address in Hex and HH = Value in Hex. Refer to **Table A-2**, **page A-3** in Appendix for details.

The binary set-up features can also be configured from the keyboard by entering the Supervisor Set-Up mode SHIFT CTRL SETUP and then entering the command sequence CTRLB. The cursor control arrow keys are utilized to address the set-up byte, locate the bit to be configured, and to toggle the bit to 1 or 0 as desired. \uparrow is used to decrease the address character or set a bit = 1. \downarrow is used to increase the address character or set a bit = 0. \rightarrow moves the cursor right, and

OPERATOR CONVENIENCE SETUP

CLK N 🍸

Operator Convenience Set-Up functions may be displayed by entering SHIFT SET UP. The first setup function will be displayed on the status line (as above). With cursor over the Y, CLICK is enabled. To disable CLICK, press so cursor is over N. The press to display the next available fucntion... STS NRM BLANK BLINK REV RED RED-REV

...status line then appears, with REV (Reverse) as the default condition. If one of the other conditions is desired, press or a to move the cursor block over the desired condition. Press to advance to the next Set-Up function.

SUPERVISOR SETUP

CLK N 🕅

Supervisor Set-Up functions may be displayed by entering [SHIFT]CTRL [SET UP]. The first set up function will appear on the status line (line 25). The cursor block indicates default condition Y. To move cursor to N, press the \square key. When the proper function is selected, press the \square to advance to the next function.







Figure 3-2. ADM 12 Set-Up Mode Display and Associated Keys

SET UP STATUS	F 1	7 F	18 2	F 19 3	F 4	D F	21 5	F 6 6	F 723	F	24 8	F 925	F	26 10	F 27 11	F 2 1	28 2	F 29 13	F ³⁰ 14	F	31 15	F 32 16	LINE Insert Char	PAC ERA LIN	SE D	LINE IELETE Char	G EDT ISERT	BREAK
LOCAL Scape Scape	ł		@ 2	# £ 3	\$ 4		% 5	^ 6	& 7	1	3	(9		8		+=	Ţ	$\tilde{\mathbf{v}}$	BACK SPACE]	SEND PAGE NEW	SEND MSG GRAPH	PRINT REV BK GRD		TAB	CLEAR DELETE	AUX Port	SCROLI
	CAP	Q	 v	<u> </u>	E	R	T	Y		U	1		0	P		[]	}			L	LINE	NORM	NORM	╵┝	7	8	9	_
CTRL	LOCK		z	s x	D	F C	' V	G B	H N	ר 	M			 >	; ;	Ť	SHIF	TURN	Ļ] [←	† HOME		۱H	4	5 2	6 3	, E N
L			-	 ^	'							<u> </u>	<u> </u>	•	/	1			\][╵┝		1		N T E R
L]						I							<u> </u>	_				ιL		LINE ↓ FEED ↓]			D		T E

Figure 3-3. Programmable Edit Keys

↑ moves the cursor left. All binary set-up features can be saved in non-volatile memory by the control sequence CTRLS while the ADM 12 is in the Supervisor Set-Up mode.

3.3 OPERATIONAL MODES

The ADM 12 provides several different modes of operation which are selectable by the host or from the keyboard. The power-on mode conditions are established by the last saved functions selected in the associated Set-Up features. The various command sequences used to change the operating characteristics of the ADM 12 are detailed in paragraph **3.6, on page 3-16**.

3.3.1 On-Line or Local Mode

On-Line Mode -- When the terminal is placed On-Line, data that is received will be displayed or acted upon, and keyboard entries will be transmitted and/or displayed, depending on the communications mode selected.

Local Mode -- In Local Mode, the terminal ignores the communications interface. Keyboard entries are displayed or acted upon locally. No data transmission takes place between the terminal and the host computer. The RTS (Request To Send) and DTR (Data Terminal Ready) signals are held low (busy).

3.3.2 Conversation or Block Mode

Conversation Mode -- When the terminal is placed in the Conversation Mode, data entered on the keyboard is immediately transmitted to the host computer, character by character, and depending upon the Duplex mode is either acted on and displayed or sent directly to the host without display.

Block Mode -- When the terminal is placed in the Block Mode, data entered on the keyboard is processed locally by the terminal and transmitted to the host computer by the various send functions that are available in the **ADM 12**. Available send functions are further discussed in **Section 3.4.7**, **page 3**-**12**.Note that the six programmable edit keys (shiftable to twelve) can be programmed to transmit directly to the host or to act locally independent of the communications mode of the terminal. Refer to **Section 3.4.8**, **page 3**- 13 for details on the use of the programmable edit keys.

3.3.3 Duplex Operation

The Duplex Mode selected in Set-Up or through escape sequence code controls the general handling of the RS-232C signal RTS (Request To Send) and, in Conversation Mode, controls the processing of key entries.

Full Duplex -- RTS is always high as long as terminal is on line. Characters transmitted to the host computer are not directly displayed on the CRT. In order for data to be displayed, it must be echoed from the host computer.

Half Duplex -- RTS is raised high on attempt to transmit and dropped low 70 to 140 milliseconds after a "CR" (0 D Hex) or "End of Block 1" character selected in set-up is transmitted. Characters that are entered from the keyboard are transmitted to the host computer and are also routed to the CRT to be displayed and acted upon locally.

3.3.4 Program Mode

Setting Program Mode via the ESC U key sequence causes the terminal to display all 128 ASCII characters. Normally, the 32 control codes (00-1F) cause a particular action to be performed. However, in Program Mode, the control codes are displayed instead of being acted upon. This allows the embedding of formatting information that is particularly useful in print output and as a diagnostic aid. The ESC X and/or ESC u key sequence causes the terminal to exit Program Mode. The terminal will not respond to "X-ON" or "X-OFF" while in the Program Mode. Host control of Program Mode can be disabled through Set-Up.

3.4 KEYBOARD OPERATION

The operator uses a keyboard very similar to that of a standard office typewriter to enter data and perform control operations. Functionally, the keyboard consists of the displayable 96 ASCII character set keys and various control or modifier keys. All keys are auto-repeating at the rate of 15 characters per second, except the Edit keys, Function keys and, Special Operation keys: Insert Mode, Page, Clear, Background, Graphics, Back



Figure 3-4. ADM 12 Standard Keyboard Operation Characteristics



Figure 3-4. ADM 12 Standard Keyboard Operation Characteristics (continued)



Figure 3-4. ADM 12 Standard Keyboard Operation Characteristics (continued)

Tab, Tab, Page Edit, Aux. Port, and Break. Auto repeat starts after a key has been depressed for a minimum of one second. Figure 3-4, page 3-7 describes the functions of the ADM 12 keyboard. Paragraphs 3.4.2 through 3.4.9 detail the keyboard by the following types of operation.

- Alphanumeric and Punctuation Keys
- Numeric Keypad
- Modifier Keys
- Cursor Control Keys
- Edit Keys
- Transmission Keys
- Programmable Function Keys and Edit Keys
- Special Operation Keys

3.4.1 Keystroke Entry

Ordinarily, each keystroke that is typed by the operator is independent of the one preceding or following it. However, the keyboard is scanned in such a manner that even a very fast typist will not be able to overrun the keyboard. This is accomplished by the 3-key rollover data entry protection, plus a 32 keystroke buffer, provided by the **ADM 12**. The keyboard also provides audible feedback by a user-selectable key-click feature that indicates a valid key closure.

Keys such as <u>SHIFT</u> and <u>CTRL</u> are used in conjunction with other keys to modify their operation.

3.4.2 Alphanumeric and Punctuation Keys

The upper/lower case alphabet, numerics, and punctuation characters from the 96 displayable ASCII character set, plus space and DEL are displayed and/or transmitted (depending upon the communication mode selected) when a key is pressed. Keys that have a double legend produce the lower-case or lower legend when unshifted, and the upper-case or upper legend when pressed with SPACE SHIFT kev. The the kev generates an ASCII 20 Hex code for transmission and occupies one space on the display screen. The DEL key produces an ASCII 7F Hex code for transmission, but will not occupy a space on the display screen unless Program Mode is set. [DEL] may be used by the host computer as a character erase code.

3.4.3 Numeric Keypad

The ASCII numerals \emptyset through 9 and minus, comma, and period are displayed and/or transmitted when a key is pressed. The ASCII hex codes generated are identical to the lower legend numerals and punctuation on the main keyboard area.

3.4.4 Modifier Keys

The following keys do not generate any output by themselves, but modify the code generated by the alphanumeric keys on the keyboard.

SHIFT causes the upper legend character of the double legend key to be produced when pressed in conjunction with either **SHIFT** key. The 26-alpha characters are shifted for upper-case, and unshifted for lower case.

CAP LOCK this alternate action key causes the 26-alpha characters to stay shifted (uppercase) when on. The numeric and punctuation keys, as well as the various control keys, are not affected.

CTRL the control key causes one of the 32-ASCII control codes to be generated when pressed in conjunction with an otherwise displayable character key. The character generated will not display a character on the display unless Program Mode is set. Some control codes are utilized by the **ADM 12**, refer to **Table 3-2**, **page 3-17** for details.

3.4.5 Cursor Control Keys

The Cursor is used to indicate the next character position to be entered on the display. The cursor may be positioned by remote commands from the host computer or by cursor control keys on the keyboard. The following keys are used to position the cursor on the display:

 $\uparrow \lor \leftarrow \Rightarrow$ and HOME move the cursor as indicated and transmit the ASCII control codes listed in Table 3-2, page 3-17.

Either [TAB] key may be used to generate a horizontal tab code (HT) and depending upon modes of the terminal will be acted upon moving the cursor right to the next horizontal tab setting or right margin and to beginning or end of a protected field. [SHIFT[TAB] is a Back Tab and will generate an ESC I code and depending upon modes of the terminal will be acted upon moving the cursor left to the next horizontal tab setting or left margin.

BACKSPACE causes the cursor to move one character position to the left. A BS code is generated. This key can be reprogrammed in Personality Mode of Set-Up.

<u>RETURN</u> moves the cursor to the beginning of the line it was in, or the beginning of the next line depending upon the Personality character selection mode. A Carriage Return Code (CR) is generated by the Return key. If the Personality selection is made for a line feed as well as carriage return when RETURN key is depressed, the Line Feed Code (LF) is generated after the Carriage Return Code (CR).

The ENTER key, on the numeric keypad is programmable through Set-Up Mode. The operator can program the ENTER key to any one or two codes as desired. The default function is carriage return.

LINE FEED moves the cursor to same column position on the next line down unless protected. This will cause a scroll operation to occur if the cursor was on the bottom line of the scrolling region and the scroll operation is enabled. A LF code is generated.

3.4.6 Edit Keys

There are six edit keys on the ADM 12. Three, [INSERT CHAR/LINE],

ERASE LINE/PAGE and DELETE CHAR/LINE, are also programmable along with the transmission keys (figure 3-3, page 3-5). The programming functions are discussed in Section 3.4.8, page 3-13. In this section only the default function conditions which relate to the key legends will be explained. As set from the factory, all the edit keys are local functions and as such do not generate codes for transmission to the host. The functions can, however, be also generated by the host computer or from the keyboard by escape sequences as described in **Table 3-3**, page 3-19. Several of the functions also have the capability of certain "Personality" changes which are described in **Section 3.7.5**, page 3-33.

SHIFT CLEAR erases all characters on the screen and moves the cursor to the HOME position (first unprotected position). The erased characters are replaced with the selected Fill 1 character in set-up.

DELETE transmits DEL code 7F hex.

ERASE-LINE erases all unprotected characters from the cursor position through to the next protected location or the end of the line. The erased characters are replaced with the character selected in Fill 1. The cursor position does not change.

SHIFT ERASE is ERASE-PAGE and will erase all unprotected characters from the cursor position through the end of the current page. The erased characters are replaced with spaces (Fill 1). The cursor position does not change.

INSERT-CHAR moves the character under the cursor and all following characters in the line one position to the right up to the right margin or next protected field. Wraparound will occur only if Page Edit mode is selected.

SHIFT INSERT is INSERT-LINE and moves the entire line containing the cursor as well as following lines down one line. The new line is filled with the character Fill 1 (default space) selected during set-up. Last line in scrolling region is lost. If Protect Mode is on or cursor is positioned outside of scrolling region, this command is ignored and an audible tone will sound.

SHIFT DELETE is DELETE-LINE and the line containing the cursor will be deleted as long as Protect Mode is off. The following lines in the scrolling region below the cursor will be moved up one line and a line containing the Fill 1 character (default space) will be added as the last line in scrolling region as long as cursor is within scrolling region. If Protect Mode is on or cursor is positioned outside of scrolling region, this command is ignored and an audible tone will sound.

[INSERT MODE] places the terminal in the Insert Mode which causes all characters received to be inserted at the cursor position. The character under the cursor and all characters to right will move one position to the right as characters are received. Wraparound will occur if terminal is also in Page Edit mode. SHIFT results in PG EDT which is Page Edit mode of the terminal whereby characters entered or deleted at the cursor position will result in wraparound of lines below cursor. Page Edit is inoperative in Embedded Attributes Mode.

3.4.7 Transmission Keys

There are three transmission keys in the **ADM 12** which are also programmable as described in **Section 3.4.8**, page 3-13. This section describes the default function conditions which relate directly to the key legends. These three transmission keys are: [SEND], [SEND MSG] and [PRINT].

SEND is Send Page Foreground, (ESC 5), which causes all data, excluding nulls, to be transmitted from the top left corner of the page up to, and including, the cursor. Protected fields are replaced with a single FS character selected in set-up unless selected as a NULL. Each line will be limited on the right by the Right Margin, and except for the last line of the transmission, each line will be terminated by the two New Line Characters selected in Set-Up. If Set-Up byte 3, bit 4 ('Inhibit New Lines In Send') is set to '1', or if the New Line Characters are programmed as nulls, they shall be omitted. The transmission will be terminated by the two End of Block Characters selected in Set-Up unless they are nulls.

SHIFT SEND is Send Line Foreground, (ESC 4), which causes all data, excluding nulls, to be transmitted starting at column 1 of the line containing the cursor up to, and including, the cursor. Protected fields replaced by one FS character selected in Set-Up unless it is a null. New line characters will be omitted and the transmission will be terminated with the two End Of Block Characters selected in Set-Up.

SEND MSG is Send Message Foreground, (ESC S), which causes all data from the character following the previous STX (Ø2 Hex) embedded on the screen (or column 1 row 1 of current page if no STX is embedded) up to the character prior to the next ETX (Ø3 Hex) embedded on the screen (or the character at the right margin on the last row of the current page if no ETX is found).

All lines will be bound on the right by the Right Margin selected, and except for the last line of the transmission, will be terminated with the two New Line Characters selected in Set-Up (unless they are nulls). The last line will be terminated with the two End Of Block Characters selected in Set-Up unless they are nulls. All protected fields will be replaced with a single FS character selected in Set-Up unless it is a null.

SHIFT SEND MSG is Send Message All, (ESC s), which causes all data from the character following the next previous STX (02 Hex) embedded on the screen (or column 1, row 1 of the current page if no STX embedded) up to the character prior to the next ETX (03 Hex) embedded on the screen (or the character at the right margin on the last row of the current page if no ETX is found.)

All lines will be bound on the right by the Right Margin selected, and except for the last line of transmission, will be terminated with the two New Line Characters selected in Set-Up (unless they are nulls). The last line will be terminated with the two End Of Block Characters selected in Set-Up unless they are nulls. All protected fields will be delineated with ESC) and ESC (. Attribute changes will be preceeded by an ESC G X.

PRINT is Print Page Formatted, (ESC P), which causes a page print starting from the upper left corner of the page and continuing up to but not including an embedded GS Stop Code (1D Hex) or the cursor, whichever occurs first. The print will be bound on the right by the Right Margin selected in Set-Up.

The PRINT shall start out with CR (ØD Hex), Null (ØØ Hex), LF (ØA Hex), Null and will
terminate each line thereafter (until the stopping criteria above is met) with the same code. Trailing blanks will be suppressed and all nulls will be suppressed. Protected fields will be replaced with spaces if in Set-Up Print Protected As Spaces is set. Embedded attributes will be replaced with spaces regardless of protection if in Set-Up Print Attribute As Space is set.

SHIFT PRINT is Print Page Unformatted, (ESC P), which causes a page print starting from the upper left corner of the page and print up to but not including an embedded GS Stop Code (1D Hex) or the cursor, whichever occurs first. The print shall be bound on the right by the Right Margin selected in Set-Up.

The ADM 12 has two additional send commands that can either be used by escape sequences or alternately by programming one or more of the programmable keys as described in **paragraph 3.4.8, below**. The operation of these send functions is described here for reference:

SEND PAGE ALL (ESC 7) causes all data, excluding nulls, to be transmitted from column 1, row 1 of the current page up to and including the cursor. Embedded attributes or attribute changes (non-embedded mode) will be delineated by 'ESC G X' where X is the ASCII encoded character which defines the attribute of the next field. Protected attributes will be preceeded by 'ESC)' and terminated by 'ESC ('. Each line is limited on the right by the Right Margin, and except for the last line of the transmission, each line shall be terminated by the two New Line Characters selected in Set-Up. If Set-Up byte 3, bit 4 ('Inhibit New Lines In Send') is set to '1', or if the New Line Characters are programmed as nulls, they shall be omitted. The transmission shall be terminated by the two End Of Block Characters selected in Set-Up unless they are nulls.

SEND LINE ALL (ESC 6) causes all data, excluding nulls, to be transmitted from column 1 of the line containing the cursor up to, and including the cursor. Embedded attributes or attribute changes (nonembedded mode) will be delineated by 'ESC G X' where X is the ASCII encoded character which defines the attribute of the next field. Protected attributes are preceeded by 'ESC)' and terminated by 'ESC ('. The transmission will be terminated by the two End Of Block Characters selected in Set-Up. If either End Of Block Character is a null, it shall be omitted.

3.4.8 Programmable Function Keys and Edit Keys

The ADM 12 has 16 function keys, giving the user 32 programmable function sequences with the SHIFT key. In addition, there are 6 Edit/tranmission keys that are user programmable for an additional 12 function sequences with the SHIFT key. The Edit/transmission keys that are programmable are: [SEND], [SEND MSG], **INSERT CHAR/LINE** PRINT. ERASE LINE/PAGE, and DELETE CHAR/LINE . Thus the ADM 12 has up to 44 functional sequences that are user programmable. All of the programmable keys have an independent destination code, Local or Host, and four characters each of program space. In addition, all programmed functions can be saved in non-volatile memory through the Set-Up mode.

The function sequences may be programmed from the host with 4 ASCII characters or from the keyboard locally with 4 "keystrokes". Keystrokes may be ASCII characters 00-7F Hex or local action keys like PAGE ERASE. It's even possible to link function sequences into longer sequences by programming the 4th keystroke with the actual function key you wish to link to. This also enables looping if you link a function on itself. A looping function will repeat until the operator presses SHIFT and BREAK keys simultaeously.

All the programmable keys are programmed via the escape sequence $[\underline{ESC}][!]X[P_0]P_1$

 P_2 P_3 P_4 where "X" is an ASCII character corresponding to the key to be programmed, P₀ is the destination code "K" (4B Hex) for Local "H" (48 Hex) for Host, and P₁-P₄ are the actual codes to be programmed into the function sequence (see **figure 3-5**, **page 3-14**). In each sequence all four characters, P₁-P₄, must be entered. NULLS

FUNCTION KEY	X	Ρ.	Ρ 1	P 2	P 3	P₄
F1	@	н	ѕон	@	CR	NULL
F2	А	н	SOH	А	CR	NULL
F3	В	н	SOH	В	CR	NULL
F4	с	н	SOH	С	CR	NULL
F5	D	н	SOH	D	CR	NULL
F6	E	н	SOH	E	CR	NULL
F7	F	н	SOH	F	CR	NULL
F8	G	н	SOH	G	CR	NULL
F9	н	Н	SOH	н	CR	NULL
F10	1	н	SOH	I	CR	NULL
F11	J	н	SOH	J	CR	NULL
F12	к	н	SOH	к	CR	NULL
F13 F14	M	H H	SOH SOH	L M	CR CR	NULL NULL
F15 F16	N O	н Н	SOH SOH	N O	CR CR	NULL NULL
F17 F18	P Q	H H	SOH SOH	P Q	CR CR	NULL NULL
F19 F20	R S	H H	SOH SOH	R S	CR CR	NULL NULL
F21 F22	T U	H H	SOH SOH	T U	CR CR	NULL NULL
F23 F24 WITH	v w	H H	SOH SOH	V W	CR CR	NULL NULL
F25 F26	X Y	н н	SOH SOH	X Y	CR CR	NULL NULL
F27 F28	Z [H H	SOH SOH	Z [CR CR	NULL NULL
F29 F30	1	H H	SOH SOH	1	CR CR	NULL NULL
F31 F32	<u>^</u>	H H	SOH SOH	<u>^</u>	CR CR	NULL NULL
SEND SHIFT SEND	` a	K K	ESC ESC	5 4	NULL NULL	NULL NULL
SEND MSG SHIFT SEND MSG	b c	к к	ESC ESC	5 s	NULL NULL	NULL NULL
PRINT SHIFT PRINT	d e	к к	ESC ESC	P p	NULL NULL	NULL NULL
INSERT CHAR SHIFT INSERT LINE	f g	к к	ESC ESC	Q E	NULL NULL	NULL NULL
ERASE LINE SHIFT ERASE PAGE	h i	к К	ESC ESC	T Y	NULL NULL	NULL NULL
DELETE CHAR SHIFT DELETE LINE	j • k	к к	ESC ESC	W R	NULL NULL	NULL NULL

Figure 3-5. Factory Set Default Condition of Programmable Keys

(CTRL @) programmed into a function sequence are not transmitted. A function can be filled partially with data then padded with NULLS or totally disabled by completely filling it with NULLS. The factory set default conditions of the programmable keys are listed as follows with the ASCII codes "X" and P $_0$ for each key.

Note

While programming the function keys and edit keys, the codes that you type in as well as the data that you are programming into that key will not be displayed on the screen.

When the programmable function sequences are selected, they are stored in temporary "working" memory and will be lost if the terminal is powered-down. To cause the programmed functions to be saved in nonvolatile memory it is necessary to enter the Set-Up mode and perform a <u>CTRL</u> as described in Section 3.2.2, page 3-2. The functions defined for each key will then be retained after the terminal is powered off. While in the Set-Up mode, <u>CTRL</u> will cause the terminal to set the programmable functions to the default value, and <u>CTRL</u> and causes the terminal to restore the values currently in non-volatile memory.

3.4.9 Special Operation Keys

The keys listed below perform special operations or have a unique effect on the **ADM 12**.

The <u>SET UP</u> key causes the **ADM 12** to enter Operator Convenience Set-Up when pressed in conjunction with the <u>SHIFT</u> key.

The <u>SET UP</u> key causes the ADM 12 to enter Supervisor Set-Up Mode when pressed in conjunction with <u>SHIFT</u> and <u>CTRL</u>, as described in paragraph 3.2, page 3-1.

STATUS key causes the **ADM 12** to exit Set-Up Mode when pressed, as described in **paragraph 3.2, page 3-1**. Also brings the Status Line back to the 25th line and acknowledges a parity error indicated (PE). <u>SHIFT CTRL</u> key determines the display of the 25th line, toggling between the available functions each time it is depressed. The available functions of the 25th line are: Terminal Status, Host Message 1, Host Message 2, and OFF.

BREAK causes a break (mark) condition to be presented on the transmit data line of the Primary (Modem) Port for approximately 600 milliseconds. This operation has no effect on the terminal and is usually used for control signaling to the host device.

SHIFT BREAK causes the terminal to abort prints, sends, or looping Function sequences. The keyboard buffer will also be purged.

SHIFT CTRL BREAK causes the terminal to perform a power-up reset.

ESC generates a special control code (Hex 1B), which is used for command operations and usually followed by one or more characters. Thus, escape is usually considered a "Lead-In" character for terminal control operations. Paragraph 3.6.2, page 3-16 and Table 3-3, page 3-19 detail the escape sequences utilized by the ADM 12. Pressing SHIFT with ESC key causes a local escape function to be processed i.e., the control operation will be acted upon by the terminal but the code will not be transmitted.

NEW LINE causes the ADM 12 to generate a new line code US (1F Hex). Pressing SHIFT with the NEW LINE key generates NEW PAGE and causes the ADM 12 to switch to the alternate display page. BKGND-NORM causes the terminal to return to normal (dark) display background from Full Screen Reverse Video mode. With SHIFT, this key is BKGND-REV which causes the ADM 12 to enter the Full Screen Reverse mode.

NORM is used to return the display mode to normal from Business Graphics display mode; with SHIFT, this key is SHIFT GRAPH and causes the ADM 12 to enter the Business Graphics display mode. **SCROLL** causes the ADM 12 to toggle between Scroll Enable and Scroll Disable. In Scroll Disable the ADM 12 will discontinue the scrolling operation on the display and X-OFF will be generated provided the X-ON/X-OFF handshaking has been selected. If DTR handshaking is enabled then DTR will drop when scrolling is disabled and then raise again if <u>SCROLL</u> is pressed again.

3.5 DISPLAY CHARACTER FORMAT

The standard ADM 12 character set contains 128 ASCII characters, 32 of which are control characters. The entire character set may be displayed on the CRT screen by setting Program mode in the unit or by preceding a control character with an ESC code. Wraparound is automatically enabled when Program mode is selected. The key sequence ESC U selects Program mode; ESC X or ESC u exits Program mode. The displayable USASCII character set and control codes are shown in Appendix A.

In addition to the standard ASCII character set, the ADM 12 provides a business graphics capability. The business (limited) graphics operation codes are shown in Table 3-4, page 3-39, and the business graphics character set is illustrated in Appendix A, Table A-8, page A-13.

3.5.1 International Fonts

As an option to the **ADM 12**, international fonts and keyboards for the following languages are available:

- United Kingdom
- German
- French
- Spanish
- Swedish/Finnish
- Danish
- Norwegian

International keyboard layouts are shown in the **Appendix** for reference.

3.6 COMMAND CODE SET

3.6.1 Control Codes

The operational characteristics of the ADM 12 are controlled, in part, by a group of control codes which may originate from the host or be generated from the keyboard by pressing the <u>CTRL</u> key simultaneously with another key as defined in **Table 3-2**, **page 3-17**. Control codes can be displayed when Program Mode is set. When Program Mode is set, the ADM 12 will display but not act upon the recognized control code. <u>ESC[U]</u> sets Program Mode and <u>ESC[X]</u> or <u>ESC[u]</u> exits Program Mode.

3.6.2 Escape Sequences

An Escape sequence is formed by the combination of the "escape" ASCII control code, followed by one or more other ASCII characters. Each Escape sequence controls a specific terminal operation. Some operations are one-time only, others remain operative for as long as power to the unit is not interrupted or until terminated by a control code or another escape sequence.

Table 3-3, page 3-19 shows the escape sequences used by the ADM 12, whether initiated from the host computer or the keyboard. Escape sequences for the ADM 12 are also summarized in a chart on page A-7 of the Appendix for reference.

Local Escape <u>SHIFT ESC</u> – Pressing <u>SHIFT</u> in conjunction with the <u>ESC</u> key causes the escape sequence in its entirety to be acted upon locally and not transmitted.

Table 3-2. Control Codes Utilized by the ADM 12

OPERATION	FROM HOST (ASCII Code)	FROM KEYBOARD	DESCRIPTION
Answerback	ENQ	CONTROLE	Transmits 32 byte answerback sequence (nulls are not transmitted).
Bell	BEL	CTRL G	Sounds the audible bell tone.
Backspace	BS	← BACKSPACE CTRL H	Moves the cursor to the left to the next unprotected location. If the cursor is at the left margin, it moves up to the end of the previous line until a non- protected location is found or the cursor reaches the Home position.
Forespace	FF		Moves the cursor to the right to next unprotected location. If the cursor is at the right margin, it moves down to the start of the next line until a non- protected location is found or the cursor reaches the lower right corner of the scrolling region.
Linefeed	LF	LINEFEED CTRL J	Moves the cursor to the next line down in same column. If the intended location is protected, the cursor will move right to the first unprotected location or stop at the lower right corner of the scrolling region if an unprotected character can- not be found. If the cursor is located on last line, a scroll operation will be performed when Write Protect is off and if scrolling is enabled in set-up.
Upline	VT	1 CTRL K	Moves the cursor up to the previous line in same column. If the intended location is protected, the cursor will continue from right to left to the first unprotected location stopping at Home position if an unprotected location cannot be found.
Carriage Return	CR	RETURN CTRL M	Moves the cursor to the left margin or to the first unprotected character position in current line. If none are found, cursor will continue in fol- lowing lines stopping at unprotected position or, if none found it will shop at the lower right of scrolling region.
Home Cursor	RS		Moves the cursor to the first unprotected location in the scrolling region.

Table 3-2. Control Codes Utilized by the ADM 12 (continued)

OPERATION	FROM HOST (ASCII Code)	FROM KEYBOARD	DESCRIPTION
Tab	ΗT	TAB CTRL I	Moves the cursor to the next horizontal tab setting on current line or to the right margin if no tabs are set. If Write Protect mode is on tabs are disabled and cursor will move to start of next unprotected field or stop at lower right of scrolling region if unprotected fields are not found.
Escape	ESC	ESC CTRL [Recognized by the terminal as a code extension character which must be followed by otherwise displayable character(s) to invoke a specific terminal operation.
X-ON Character	DC1	CTRL Q	Transmits the default X-ON character (DC1) if no other X-ON character is selected in Set-Up Mode.
X-OFF Character	DC3	CTRL S	Transmits the default X-OFF character (DC3) if no other X-OFF character is selected in Set-Up Mode.
Reset Print	DC4	CTRL T	Resets copy print mode and/or print transparent mode.

OPERATION	SEQUENCE	DESCRIPTION
Program Function Keys	ESC ! X P ₀ P ₁ P ₂ P ₃ P ₄	Program function key, where X is ASCII encoded character (@ thru k), P_0 is destination code, P_1 - P_4 string of characters to be programmed. Refer to Section 3.4.8, page 3-13 .
Keyboard Enable	ESC "	Unlocks the keyboard unless keyboard buffer is full. Host execution only.
Keyboard Disable	ESC #	Locks keyboard and disables keyclick. Host execution only. Entering Set-Up mode will temporarily unlock keyboard until Set-Up mode is exited. SHIFT CTRLIX in Set-Up mode will unconditionally unlock keyboard.
Set Protect Mode	ESC &	Enables Protect mode and all characters on screen with write protect set shall be protected from overwrite.
Clear Protect Mode	ESC '	Disables Protect mode and all characters on the screen are unprotected.
Set Write Protect Mode	ESC)	Enables Write Protect mode and all characters written to the screen are write protected and displayed with the 'PRTMSK' attribute selected in Set-Up mode.
Clear Write Protect Mode	ESC (Disables the Write Protect mode and all characters written to screen, with exception of embedded attributes, will be unprotected and displayed in the current video attribute.
Set Block Mode	ESC B	Terminal enters Block mode of operation. Refer to Section 3.8.2, page 3-34.
Set Conversation Mode	ESC C	Terminal enters Conversational mode of operation. Refer to Section 3.8.1, page 3-34 .
Set Page Edit Mode	ESC N	Terminal enters Page Edit mode of operation. Characters entered or deleted will cause wraparound on the lines below the cursor. Refer to Section 3.4.6 , page 3-11 . Inoperative in embedded attributes mode.
Clear Page Edit Mode	ESC O	Page Edit mode will be exited and Line Edit will be entered.
Set Program Mode	ESC U	The Program mode is entered where all 128 ASCII characters can be displayed. Control codes are not acted upon in this mode. Refer to Section 3.3.4 , page 3- 6 .

Table 3-3. ADM 12 Escape Sequence

OPERATION	SEQUENCE	DESCRIPTION
Clear Program Mode	ESC X or ESC u	Terminal will exit the Program mode.
Auxiliary Control	ESC A 'X'	The Auxiliary port functions will be controlled as function of "X" as follows: "0" terminate print extention mode, "1" transparent print, "2" display and print.
Print Protected as Spaces	ESC @	Screen data will be sent to auxiliary port and all protected characters will be sent as spaces.
Send Line Foreground	ESC 4	All data except nulls will be transmitted starting at column 1 of line containing cursor up to and including cursor. Protected fields will be replaced by single FS character. Refer to Section 3.4.7, page 3-12.
Send Page Foreground	ESC 5	All data except nulls will be transmitted starting at column 1, row 1 up to and including cursor. Protected fields will be replaced by a single FS character. Refer to Section 3.4.7, page 3-12.
Send Line ALL	ESC 6	All data except nulls will be transmitted starting at column 1 of line containing cursor up to and including cursor. Write protected data will be delineated by ESC) and ESC (. Embedded attributes or attribute changes will be delineated by "ESC G X". In non-embedded mode graphics field will be delineated by "ESC L C" and "ESC L A" Refer to Section 3.4.7, page 3-12.
Send Page ALL	ESC 7	All data except nulls will be transmitted starting at column 1, row 1 of the current page up to and including cursor. Write protected data will be delineated by ESC) and ESC (. Embedded attributes or attribute changes will be delineated by "ESC G X". In non-embedded mode graphics fields will be delineated by "ESC L C" and "ESC L A". Refer to Section 3.4.7, page 3-12.
Set Insert Mode	ESC q	Terminal enters Insert mode where characters received will be inserted at cursor position and all characters under and to the right of cursor will be moved one position to the right. Wraparound will only occur if Terminal is also in Page Edit mode. Refer to Section 3.4.6, page 3-11.
Clear Insert Mode	ESC r	Terminal will exit the Insert mode of operation.

OPERATION	SEQUENCE	DESCRIPTION
Insert Character	ESC Q	A space shall be inserted at current cursor position and all characters in same line to right of and under cursor moved one location to right. Wraparound will occur if Page Edit mode is set.
Delete Character	ESC W	The character under the cursor will be deleted and all unprotected characters in same line moved one position to left. Wraparound will occur if Page Edit mode is set.
Clear All Modes	ESC Z	The following modes will be cleared simultaneously: Page Edit Mode, Protect Mode, Write Protect Mode, Insert Mode, Smooth Scroll Mode, Scrolling Region Mode. The video will also be restored.
Clear All to Fill 1	ESC +	All characters on current page will be replaced with character selected in Set-Up mode as Fill 1 (Default to Spaces). The cursor will return to Home position, Write Protect mode, Protect mode, and Scrolling Region will be reset.
Clear All to Fill 2	ESC *	All characters on current page will be replaced with character selected in Set-Up mode as Fill 2 (Default to Nulls). The cursor will return to Protect mode, and Scrolling Region will be reset.
Clear All to Protected Spaces	ESC ,	All characters on current page will be replaced with character selected in Set-Up mode as Fill 1 with Write Protect set. The cursor will return to Home position and all modes will be reset.
Clear Foreground to Fill 1	ESC ;	If in Set-Up mode Clear Command Control set to 1, this functions same as ESC +. If Clear Command Control set to 0, unprotected only characters will be replaced with Fill 1 and Write Protect mode is reset. Cursor returns to Home position or first unprotected location from Home position.
Clear Foreground to Fill 2	ESC :	If in Set-Up mode Clear Command Control set to 1, this functions same as ESC *. If Clear Command Control set to 0, unprotected only characters will be replaced with Fill 2 and Write Protect mode is reset. Cursor return to Home position or first unprotected location from Home position.
Clear Typewriter Tabs	ESC 0	All typewriter tab stops will be cleared.

OPERATION	SEQUENCE	DESCRIPTION
Set Tab	ESC 1	A tab stop will be set at the current cursor horizontal value.
Clear Tab	ESC 2	If tab stop exists at current cursor horizontal position it will be cleared.
Address Cursor Page, Row, Column	ESC - PRC	The cursor shall be moved to the page, row and column described by ASCII encoded parameters "P", "R" and "C". Refer to Section 3.9.2, page 3-34 .
Read Cursor Page, Row, Column	ESC / PRC	The terminal will send the host the ASCII encoded cursor position "P", "R" and "C" terminated with a "CR". Refer to Section 3.9.2, page 3-34 .
Address Cursor Current Page	ESC = RC	The cursor will be moved to the row and column described by the ASCII encoded parameter "R" and "C" for the current page displayed.
Read Cursor Current Page	ESC ? RC	The terminal will send the host the ASCII encoded cursor position "R" and "C" terminated with a "CR".
Write Start of Message (SOM) operation.	ESC 8	The Start of Message character STX (02 Hex) will be embedded in the display for Send Message
Write End of Message (EOM) operation.	ESC 9	The End of Message character ETX (03 Hex) will be embedded in the display for Send Message
Send Message Foreground	ESC S	All unprotected characters, excluding nulls, from the next previous STX code embedded on the screen (or column 1 row 1) up to the next ETX code embedded on the screen (or end of page) will be transmitted. All lines will be limited on the right by the user programmable Right Margin selectable in Set-Up mode. Protected fields will be replaced with a single FS character selected in Set-Up mode. All lines with the exception of the last line will be terminated by the
	an a	two NEWLINE fill characters selected in Set-Up excluding nulls unless the Set-Up feature "INHIBIT NEWLINES IN SEND" is selected. The last line will be terminated by the two END OF BLOCK fill characters selected in Set-Up excluding nulls.

OPERATION	SEQUENCE	DESCRIPTION
Send Message All	ESC s	All unprotected characters, excluding nulls, from the next previous STX code embedded on the screen (or column 1 row 1) up to the next ETX code embedded on the screen (or end of page) will be transmitted. All lines will be limited on the right by the user programmable Right Margin selectable in Set-Up mode. Attribute changes or embedded attributes will be delineated by ESC G X when X is the appropriate attribute select code. Write Protected fields will be bracketed by ESC) and ESC (and Graphics fields in Non-Embedded mode will be bracketed by ESC L C and ESC L A. All lines with the exception of the last line will be terminated by the two NEWLINE fill characters selected in Set-Up excluding nulls unless the Set-Up feature "INHIBIT NEWLINES IN SEND" is selected. The last line will be terminated by the two END OF BLOCK fill characters selected in Set-Up excluding nulls.
Erase Line to Fill 1	ESC T	All characters in the line containing cursor will be replaced with the character Fill 1 selected in Set-Up beginning at character under cursor. Erase Line will terminate at the first protected field or right margin.
Erase Line to Fill 2	ESC t	All characters in the line containing cursor will be replaced with the character Fill 2 selected in Set-Up beginning at character under cursor. Erase Line will terminate at the first protected field or right margin.
Erase Page to Fill 1	ESC Y	All unprotected characters starting from under cursor will be replaced with the character Fill 1 selected in Set-Up up to the last character on page.
Erase Page to Fill 2	ESC y	All unprotected characters starting from under cursor will be replaced with the character Fill 2 selected in Set-Up up to last character on page.
Insert Line	ESC E	If Protect mode is off and if the cursor is within the scrolling region, a line containing the Fill 1 character selected in Set-Up will be inserted in line position containing cursor. Original line will be moved down one line. Last line in scrolling region will be lost. The cursor will return to the left margin.
Delete Line	ESC R	If Protect mode is off and if the cursor is within the scrolling region, line containing cursor will be deleted and all lines below cursor will be moved up one line. A line containing Fill 1 character selected in set-up will be inserted at the bottom of the scrolling region. The cursor will return to the left margin.

OPERATION	SEQUENCE	DESCRIPTION
Formatted Print	ESC P	A "GS" code will be written at the cursor and the page print will start from column 1, row 1 of the current page and terminate at the position right before the first "GS" code on the screen. The print will be formatted and the GS code stopped at will be written over with a space.
Unformatted Print	ESC p	The page print will start from the upper left corner of the scrolling region and continue up to an embedded "GS" stop code or the cursor. Print shall be unformatted. Refer to Section 3.4.7, page 3-12 .
Print Line Formatted	ESC z	A "GS" code will be written at the cursor and the page print will start from column 1 of the current line and terminate at the position right before the first "GS" code on the line. The print will be formatted and the GS code stopped at will be written over with a space.
Back Tab	ESC I	If terminal is not in Protect mode, cursor will move back on current line stopping at the next Tab Stop or beginning of line. Refer to Section 3.9.3, page 3-35 .
Horizontal Tab	ESC i	If terminal is not in Protect mode, the cursor will move forward on current line stopping at the next Tab Stop or end of line. Refer to Section 3.9.3, page 3- 35 .
Page Back	ESC J	If the terminal is set for 2 pages of display, this sequence will toggle the page displayed.
Page Forward	ESC K	If the terminal is set for 2 pages of display, this sequence will toggle the page displayed.
Set Graphics Mode	ESC L 'X'	If the terminal is in Non-Embedded Attributes mode, this command will turn on Business Graphics mode if X = 'C' and turn it off if X = 'A'.
Set Long Page	ESC v	The page configuration will be set to one page 48 lines by 80 columns.
Clear Long Page	ESC w	The page configuration will be set to two pages 24 lines by 80 columns.
Change Personality	ESC.A XX	Causes personality parameter "A" to be set to the hex value "XX".

OPERATION	SEQUENCE	DESCRIPTION
Attribute Control	ESC G'X'	Sets the visual attributes for both Embedded Attribute mode and Non-Embedded Attribute mode. The character "X" is an ASCII encoded variable to select the combinations of available visual attributes. Refer to Section 3.7.4, page 3-33.
Set Scrolling Region	ESC c T B	The Scrolling Region boundaries shall be set for the current page. "T" is the ASCII encoded line number for the top line of the scrolling region and "B" is the bottom line. Refer to Section 3.7.1, page 3-27 .
Set Smooth Scroll	ESC e	The terminal will be placed into Smooth Scroll mode where scrolling occurs at a rate of 1 scan line per vertical refresh. Refer to Section 3.7.1, page 3-27 .
Set Jump Scroll	ESC f	The terminal will be placed into Jump Scroll mode, where scrolling occurs at a rate of 1 character row per vertical refresh. Refer to Section 3.7.1, page 3- 27 .
Set Full∕Half Duplex	ESC D 'X'	The terminal will enter the Full or Half Duplex mode of operation depending on the character "X" as follows: Full Duplex X = "F" (46 Hex) Half Duplex X = "H" (44 Hex) Refer to Section 3.3.3, page 3-6 .
Select Line 25	ESC 🛆 'X'	This sequence allows the 25th display line to display terminal status or a Host Message. 'X' = 0 = Status Line 1 = Host Message 1 2 = Host Message 2 7 = OFF Refer to Section 3.4.9, page 3-15 .
Wide Page Scroll By Line	esc <	Selects line at a time horizontal scroll when in wide page.
Wide Page Scroll By Page	ESC >	Selects page at a time horizontal scroll when in wide page.
Display Host Message #1	ESC g	This sequence causes Host Message #1 to be displayed on the 25th line.
Display Host Message #2	ESC h	This sequence causes Host Message #2 to be displayed on the 25th line.
Write Message Line #1	ESC F TEXT	This sequence channels the next 80 characters on to the 25th line as the Host Message #1.
Write Multiple Characters (V1.2 and above)	ESC m 'CHAR' N	This sequence will cause the 'CHAR' to be displayed "n" times from the cursor. n = N - 20 Hex and ranges from SPACE to DEL (20 Hex - 7F Hex).

OPERATION	SEQUENCE	DESCRIPTION
Write Message Line #2	ESC I TEXT	This sequence channels the next 80 characters on to the 25th line as the Host Message #2.
Select Baud Rate	ESC ¦ X	The baud rate will be changed to the value specified by "X": 'X' = 'E' (45 Hex) 300 Baud 'F' (46 Hex) 600 Baud 'G' (47 Hex) 1200 Baud 'J' (4A Hex) 2400 Baud 'L' (4C Hex) 4800 Baud 'N' (4E Hex) 9600 Baud 'O' (4F Hex) 19200 Baud (Also selectable in Set-Up mode.)
Cursor and Video Control	ESC~X	The active cursor attributes will be changed to the value specified by "X":
		'X' = '0' (30 Hex) Cursor and Video Off '1' (31 Hex) Video On - Cursor Off '2' (32 Hex) Cursor On - Steady Block '3' (33 Hex) Blinking Block '4' (34 Hex) Cursor On - Steady Underline '5' (35 Hex) Blinking Underline
Alignment Pattern	ESC H	The scrolling region shall be reset and the current page will be filled with ASCII H (48 Hex) for alignment pattern. Protect mode and Write Protect mode are both cleared.
Terminal Reset, Default Firmware I.D. Version	ESC o ⁴ X⁴	Special functions shall be performed on the basis of the character "X":
		X = "!" (21 Hex) Terminal shall perform complete reset.
		X = "9" (39 Hex) Terminal shall display current firmware revision and report the power up self test results.
	x	X = "A" (41 Hex) All function keys shall be set to default values.
		X = "S" (53 Hex) All active set-up conditions will be set to default values.
Write Set-Up Byte	ESC { AA HH	One byte of set-up at the address "AA" (Hex) shall be set by the Hex value "HH". Refer to Section 3.2.3, page 3-2 .
Link Function Keys	ESC } X 1 X 2	This sequence is used to link functions keys together if desired. May be sent at anytime to the terminal.
		X $_1$ = ASCII character assigned to first function key.
		X ₂ = ASCII character assigned to function key being linked (see figure 3-5, page 3-14).

3.7 DISPLAY FORMAT

The ADM 12 can display up to 24 lines of data each 80 characters wide. The 25th line may be used for Terminal Status display or host messages as determined by the <u>STATUS</u> key. The Display Memory will store up to 3840 character in one of three page configurations as selected during set-up are available for operator convenience. They are:

Two Independent Displays: Each page 24 lines of data 80 characters wide. Each page having independent scrolling regions and cursor positioning.

Long page Display: This page is 48 lines of data 80 characters wide. The display area of this page is controlled by the cursor UP/DOWN keys.

Wide Page Display: This page is 24 lines of data 158 characters wide. With selectable right margin the horizontal width of this page can be set anywhere up to 158 characters, allowing configuration of 24 x 132 which is often used for financial reporting. Horizontal scrolling is controlled by the right and left cursor keys.

Figure 3-6, page 3-28 shows the various display configurations that are available in the ADM 12.

3.7.1 Vertical Scrolling

Smooth or Jump scrolling is selectable through set-up or by the Escape sequence ESC|e| for smooth and ESC|f| for Jump Scroll. Smooth scrolling is at the rate of 1 raster line per vertical refresh and Jump scrolling is at the rate of 1 row per vertical refresh. If in Smooth scroll and Handshaking protocol has not been selected, the terminal will automatically switch to Jump scroll when the input buffer reaches 75% full. As the input buffer is emptied, 25% full, the terminal will switch back to Smooth scroll.

Scrolling can be disabled/enabled utilizing the SCROLL key on the keyboard.

Split Screen Scrolling limits can be set for the current page utilizing the escape sequence

ESC C T B. "T" is ASCII encoded line number for Top line in the scrolling region, and "B" is ASCII encoded line number for Bottom line in the scrolling region.

The ASCII encoded characters "T" and "B" define the Top and Bottom Lines of the scrolling region. "T" or "B" ranges from SPACE (20 Hex) to 0 (4F Hex) corresponding to lines 1-48. No more than 23 lines may be outside the scrolling region and the page 2 scrolling region can be no greater than 24 lines.

For a given page, only one set of scrolling limits may be defined. In the two page configuration each page may have independent scrolling regions. In the Protect Mode, scrolling is disabled. However the display window can be moved up and down by the cursor control keys. If the Screen Page Display Configuration is changed, all scrolling regions are reset.

3.7.2 Horizontal Scrolling

When the Wide Page Display (24 lines by 158 characters) is configured, the terminal will scroll horizontally by moving the cursor with the Right and Left cursor keys to the right or left margins being displayed. The user may select entire page scrolling or single line scrolling through set-up.

3.7.3 Selectable Right Margin

In all page configurations, the user may specify a Right margin. The right margin specified will determine where, if enabled, wraparound will occur and also will set the right limit for prints and sends. The Right margin is selected by either escape sequence ESC { 14 X X where "X X" is the ASCII encoded character for the column location as below:

RM 1 - 158; XX = 01-9E

For Example:

RM 80; XX = 50

Alternately the right margin can be set by going into set-up mode and entering the tab set/clear mode (SHIFT CTRL T 1-80 or SHIFT CTRL V 81-158). To set the RIGHT MARGIN the user must position the blinking cursor and enter SHIFT CTRL M.



Figure 3-6. ADM 12 Display Memory Configurations



Figure 3-6. ADM 12 Display Memory Configurations (continued)

ONE PAGE - 24 lines x 158 characters/line with horizontal scrolling, plus 25th status/message line. WIDE PAGE HORIZONTAL SCROLLING ALLOWS VIEWING OF 80 COLUMNS AT A TIME OUT OF THE 158 COLUMN TOTAL. HORIZONTAL HORIZONTAL SCROLL SCROLL COL. 1 COL. 40 COL. 120 COL. 158 DSTORJ R RBKCJDJNENRLDDWKGB M HNF Z JUQWJ HU R K BENNIOD GBXDY YC KOHTI XIK KOMPD XMUYPJ NMZCH CRG NJD HUTVI KORPD FSFRKUP HOMD D DXISO IDPJIP CCG R DF FLUFFI SPRXWFSO IGT FJ S THC CKWNYCHM QUCMSZXNR A KAB HK O IN Y VINHPGUGGX T DI OVT TPF UTH HTE CRCD OU HBEFX NKJK AT UNAUZC BIE CRC HU HWI HPBLITI SNK CUCUR LHACKMR ST FLIRNZVO EH H REDVCI BOKDU UTN, YE MCN FH Y ZO S TINJHWIP VI Y PVFRJIFN NKKDU YMKLAZ ZMLGVE HUZZ HIL YS S GI INDI KURSKA K VI OT NAKLAZ ZMLGVE HUZZ HIL YS S GI INDI KURSKA K VI OT KAKLAZ ZMLGVE HUZZ HIL YS S GI INDI KURSKA K VI OT KAKLAZ ZMLGVE HUZZ HIL YS S GI INDI RUNG RYKA NFFKHNCKICAZ ZMLGVE HUZZ HIL YS S GI THV NKKDUYMSK HIKGWCHE GI HDJ NK SKA NG NAM LOG DOWNKHOK LXSON GZYCKBOO HW YGRY UKCUMORNG HUNYGYU F F MDK J CT FDKCUMOR N FTEWH CXT&PR Q KNORG POMYHKIND JYNS V FRF IP D CNIBC F F O HU R DHX UFBDB MSBK RKDAGOVKSH RHO NH TGUE BDB HS N FTEWH CJ RURGKUP DLHAYHF Z KWIXYD DH VDRNCHGKL SSON ZOU M VEJJ RKEQUCTINGSZT YDU DHCH IN BHR O MOV P ZIGMSZ X IG S IB VCS RIPHROT P S BFNJYCC B BEGGS OUTFY NOT P RONING YM NCT JUTS G WFO'T SSRJEFGRAK CUSUS J SVHT G HKT YM TC CHYFAL LG AZ JUH SGR GXX RIN FAR D CONDRENGKEN U XCSMPP I PZ WCCN IC GOT A JVHB KM RKJBBACKAK RIN FAZ ENKYF H SD XON TO CONTACT C GOT A JVHB KM RKJBBACONNRF NI CU U XCSMP I PZ WCCN IC GOT A JVHB KM RKJBBACONNRF KLBBACAGUNARF A : THC D NG XS C F UYH HXYN UOPJFF K I I VX UNINHUB I NJ IRPCU JUNCIDCIG OB UJUJOH VJAJA BHR U O H HOKUL HUHO VJAJA BHR U O HOKUL HUHO VJGOR ID KI HUHO VJGOR ID KI HUHO VJGOR ID KI HUHO U RX UJU HUHO U KNYUS GNVIC HUHO HUHO VJGOR U NJANA SANTA PG1 R01 C081 WIDE FDX DCD CTS STATUS WIDE PAGE HAS VARIABLE RIGHT FIXED POS. MARGIN FROM COLUMN 81 TO (WON'T SCROLL) COLUMN 158. FULL PAGE SCROLL ON SINGLE LINE HORIZONTAL SCROLL (SELECTABLE) IN SET-UP MODE.

LEAR SIEGLER, INC.

DATA PRODUCTS DIVISION

Figure 3-6. ADM 12 Display Memory Configurations (continued)

LEAR SIEGLER, INC. DATA PRODUCTS DIVISION



Figure 3-6. ADM 12 Display Memory Configurations (continued)

LEAR SIEGLER, INC. DATA PRODUCTS DIVISION



Figure 3-6. ADM 12 Display Memory Configurations (continued)

3.7.4 Screen Video Attributes

The ADM 12 has both Embedded and Non-Embedded Attributes. Embedded Attributes embed a write protected space at the cursor position and the visual attribute will take place on the rest of line containing the cursor up to the next embedded attribute or the end of the line. Non-Embedded Attributes are on a character per character basis. The Embedded and Non-Embedded Attributes are selected by the escape sequence ESC[G]X].

For Non-Embedded Attributes the ASCII encoded character "X" is defined as a single character ranging from \emptyset (30 Hex) to O (4F Hex), where: X = an ASCII character as below.



For Embedded Attributes the Hex ASCII encoded character "X" is defined as a single character ranging from as follows:

Bit 1 = UNDERLINE (set-up Byte 06, Bit 2 = 1) or BLANK (set-up Byte 06, Bit 2 = 0) Bit 2 = BLINK Bit 3 = REVERSE Bit 4 = BUSINESS GRAPHICS

See Table 3-4, page 3-39.

3.7.5 Personality Character Selections

The programmable personality characters may be altered by escape sequences ESC. A XX or ESC .. A XX or by depressing CTRLP on the keyboard during set-up mode.

When <u>CTRLP</u> is depressed, the terminal will enter into the "PERS" mode. The up and down arrows are used to position the display over the character to be changed. The operator will then change the fill character that is to be programmed by entering the hex equivalent of that character. The following

personality characters are available in this mode:

moue.			
BS	(BS)	08H	BACKSPACE KEY VALUE
FL1	(SPACE)	20H	FILL 1
FL2	NULL	00H	FILL 2
NL1	(CR)	0DH	NEW LINE 1
NL2	NULL	00H	NEW LINE 2
EOB1	(CR)	0DH	END OF BLOCK 1
EOB2	NULL	00H	END OF BLOCK 2
FS	FS	1CH	FIELD SEPARATOR
XON	DC1	11H	XON CHARACTER
XOFF	DC3	13H	XOFF CHARACTER
PRTATT	DLE	10H	PROTECTED ATTRIBUTE (REDUCED)
ENT1	CR	0DH	ENTER KEY CONFIGURATION 1
ENT2	NULL	00H	ENTER KEY CONFIGURATION 2

In utilizing the escape sequences ESC. A XX and ESC... A XX the personality byte "A" is loaded with the binary equivalent of the Hex-ASCII character "XX" (OO-FF) as follows:

3.8 DATA TRANSMISSION

The ADM 12 can operate in either the Conversation Mode (character-by-character) or in the Block mode for transmission of data to the host computer.

Communications are serial asynchronous, with an ASCII character format of 1 start bit, 7 data bits, odd, even, mark, space or no parity bit, and 1 stop bit. The word structure and baud rate used for transmission are selectable in Set-Up Mode. Primary (Modem) Port communications are via a bi-directional RS-232C or optional 20mA current loop interfaces detailed in Section II. Auxiliary Port communications is uni-directional via RS-232C interfacing, also detailed in Section II.

Data communications can take place whenever the **ADM 12** is placed in On-Line Mode via Set-Up Mode. When receiving data, the **ADM 12** has an X-ON/X-OFF or DTR busy indication feature (selectable in Set-Up Mode) that is used to command the host to suspend transmission to prevent data loss. Handshaking can also be selected as NONE for systems that do not support handshaking protocol.

3.8.1 Conversation Mode

Data entered on the keyboard is immediately sent to the host computer, character by character, and depending upon the Duplex mode set up, is either immediately acted on and displayed on the monitor or acted on and displayed as the characters are echoed back from the host computer. All control codes and escape sequences are sent as entered unless [SHIFT[ESC] is used for Local Escape only.

3.8.2 Block Mode

Data entered on the keyboard is processed and displayed locally by the terminal and then transmitted to the host computer by the various SENDS that are available in the **ADM 12**. Four of the SENDS are available as default conditions to two of the programmable edit keys (see **paragraph 3.4.8, page 3-13**). These keys are <u>SEND</u> and <u>SEND MSG</u>.

All six SENDS are available by escape sequences. The various SENDS are listed below along with default conditions for keyboard:

SEND	= ESC 5 = Send Page Foreground
SHIFT SEND	= ESC 4 = Send Line Foreground
SEND MSG	= ESC S = Send Message Foreground
SHIFT SEND MSG	= ESC s = Send Message All
	ESC 6 = Send Line All
	ESC 7 = Send Page All

All of the SENDS are further described in paragraph 3.4.7, page 3-12.

3.8.3 Handshaking Protocol

The ADM 12 has the ability to signal the host of a potential data loss due to the input buffer (1024 characters) being nearly full or the terminal being otherwise unable to accept data. The type of signal that is sent depends upon the Handshake Mode setting. The Handshake Mode can be set to DTR, X-ON/X-OFF, or NONE via Set-Up Mode (refer to **Table A-1, page A-1**). If NONE is selected, then no busy signal is sent to the host computer when the input buffer is full.

If DTR or X-ON/X-OFF is selected, then the

following applies: a busy signal is selected (DTR low or X-OFF). Then whenever the input buffer contains 778 characters, the appropriate busy signal is sent to the host computer. A ready signal (DTR high or X-ON) is sent to the host computer when the buffer contents are reduced to 256 characters (25% or less of the input buffer is full), as long as data flow to the host is not halted. This allows the terminal to indicate to the host computer that it is ready to receive more data. X-ON and X-OFF transmit DC1 and DC3 signals to the host computer, respectively, but, they may be changed to any other character desired via Set-Up Mode. If the terminal receives X-ON/X-OFF when DTR mode is set, then X-ON and X-OFF are ignored by the terminal.

Executing the Transparent Print function or using Display and Print with low speed printers will cause the X-ON/X-OFF commands to be used when appropriate. The Busy/Ready condition of the printer is sensed on the Auxiliary Port via the DTR signal on Pin 20.

3.9 CURSOR CONTROL

3.9.1 Relative Cursor Positioning

The Cursor may be moved to any position on the screen typically using the separate cursor control keys. Its position signifies the next character position in the display. When data is being entered, the cursor moves one position to the right on the line or to the first position in the next line each time a character is written. The operation codes and local operations required to control the cursor positions are outlined in **Table 3-2**, page 3-17.

3.9.2 Absolute Cursor Positioning/ Reading

Absolute Cursor Addressing requires commanding the cursor to a specific display location or reading out the exact cursor location. Commanding the cursor to a specific location (loading the cursor), as well as reading the cursor position, are normally executed by the host computer. The Load/Read commands are executed by means of Escape sequences in which the page, row and column locations of the cursor are expressed in ASCII characters. See Appendix, Table A-4, page A-8 for the ASCII characters assigned to each row and column locations.

Load Cursor Operation - When the Load Cursor Operation is initiated the cursor moves to the commanded position. This operation is performed by one of two escape sequences: ESC = R C for current page and ESC - P R C for either page of the Two Page configuration. Where "P" is ASCII encoded chracter for Page, "R" is ASCII encoded character for Row, and "C" is ASCII encoded character for Column to be addressed.

"P" is defined as follows:

- 1. If during set-up, Page # Designation is set to 0/1, then: "P" = 0 for page 1 and "P" = 1 for page 2.
- 2. If during set-up, Page # Designation is set to 1/2, then: "P" = 1 for page 1 and "P" = 2 for page 2.

"R" is defined as follows:

- 1. For single page display or two independent page displays "R" is ASCII encoded character Space (20 Hex) to 0 (4F Hex) for rows 1 through 48.
- 2. For wide page displays "R" is ASCII encoded character Space (20 Hex) to 0 (4F Hex) for rows 1 through 48 and columns 1 through 80. "R" is ASCII character P (50 Hex) through g (67 Hex) for rows 1 through 48 and columns 81-158.

"C" is defined as follows:

- 1. For single page displays or two independent page displays "C" is ASCII encoded character Space (20 Hex) to 0 (6F Hex) for column 1 through 80.
- 2. For wide page displays "C" is ASCII encoded character Space (20 Hex) to 0 (6F Hex) for columns 1 through 80 and repeats for column 81 through 158 (i.e. column 82 = !).

Read Cursor Operation -- The Read Cursor operation consists of the Read command usually from the host, and an immediate response from the terminal which defines the cursor position. The host executes either of the following escape sequences; ESC ? for current page and ESC / for either page of the two page configuration. The terminal responses to these escape sequences by sending to the host computer the ASCII encoded characters as required "P" for page, "R" for row, and "C" for column followed by a "CR".

3.9.3 Tab Control

Depressing either TAB control without Shift will cause an HT (09 Hex) to be generated. If Block Mode or Half Duplex Mode is on then the cursor will move right horizontally to the next Tab Stop or right margin if TAB mode is enabled. SHIFT TAB is a Back Tab and will cause an ESC I (18, 49 Hex) to be generated. The cursor will move left horizontally to the next Tab Stop of left margin if TAB mode is enabled.

Setting and deleting tabs from the keyboard are demonstrated in figure 3-7, page 3-38. How to program tabs from the host is demonstrated in Table 3-3, pages 3-19 through 3-26.

3.10 GRAPHICS

The Graphics capability of the ADM 12, another visual attribute of the unit, allows the user to draw business type forms or simple graphs to enhance the information presentation. The graphics symbols that are available for use and the associated ASCII codes are illustrated in Appendix Table A-8, page A-13. Graphics Mode may be set from the keyboard by depressing <u>SHIFT GRAPH</u> or by the appropriate Attribute escape sequence ESC G X as described in paragraph 3.7.4, page 3-33. The terminal will exit from the Graphics Mode when the key <u>NORM-GRAPH</u> is depressed.

In Non-Embedded mode, graphics is enabled by the escape sequence ESC L C and disabled by ESC L A. In Embedded mode, Graphics mode is treated as another visual attribute and is enabled by ESC G 8 and disabled by ESC G 0. See the section on Embedded Mode visual attributes.

3.11 SET-UP MODE OPERATIONS

The general operating characteristics of the ADM 12 are controlled by the user-selectable features that are displayed, one at a time, on the status line (line 25) when Set-Up Mode is entered. Both the operator and host computer have the ability to change the Set-Up functions available in the ADM 12. Selections may be saved in non-volatile memory to re-establish the same functions on the next power-on cycle or terminal reset operation. The operation of Set-Up Mode is fully described in paragraph 3.2, page 3-1.

3.12 PRINT OPERATIONS

The Auxiliary Port of the **ADM 12** is most typically connected to a serial RO (receive only) printer. The communications of data to the auxiliary device is uni-directional via an RS-232C interface. A Busy/Ready signal level is monitored for status during print operations. The print operation is controlled by the Escape sequence Copy Print Control ESC A X and the AUX PORT key on the keyboard.

The AUX PORT key enables and disables the auxiliary port, with display toggling each time the key is depressed.

The <u>SHIFT AUX PORT</u> key enables and disables the Transparent Extension Mode.

The Copy Print Control ESC A 'X' has three ASCII coded characters which determine the print status as follows:

''X'' = 0 (30 Hex) Terminates all Print Operations

" \hat{X} " = 1 (31 Hex) Enables the mode Transparent Extention Print

"X" = 2 (32 Hex) Enables the mode Display and Extention Print

There are two types of Page Print functions in the ADM 12. Page Print Formatted and Page Print Un-Formatted. Both are disabled when Transparent Extention Print or Extention Print with Display are on. PRINT is Print Page formatted (ESC P) and SHIFT PRINT is Print Page Un-Formatted (ESC p) unless the PRINT key is programmed to perform another function as described in paragraph 3.4.8, page 3-13. Print Page Formatted and Print Page Unformatted are further described in paragraph 3.4.7, page 3-12.

3.13 RESET OPERATIONS

The ADM 12 can be reset in one of two ways: a Power-On Reset, or a Reset Terminal command. Both cause the Self-Test to be executed and are detailed below.

3.13.1 Power-On Reset

A Power-On Reset consists of a complete recycling of the ADM 12 functions, including power. This is accomplished by setting the ON/OFF switch to OFF, waiting at least 10 seconds, then setting the switch to ON. All display and other volatile memory is erased when powering down. Upon power-up, the unit is subject to the complete Power Turn-On procedure specified in Section II.

3.13.2 Reset Terminal Command

At any time during operation pressing <u>SHIFT|CONTROL|BREAK</u> will perform a complete power-up reset. The **ADM 12** will perform special functions based on the escape sequence ESC o X as follows for the ASCII encoded character "X".

"X" = ! (21 Hex) the terminal will perform a complete power-up reset.

"X" = 9 (39 Hex) the terminal will display the current firmware version (VX.XX) followed by the results of a power up self test. If this request comes from host, firmware version and results of test will also be transmitted to host.

"X" = A (41 Hex) All of the Programmable Function keys will be set to the default values in paragraph 3.4.8, page 3-13.

"X" = S (53 Hex) All of the active Set-Up conditions will be set to the default values.

3.13.3 Self-Test

When the **ADM 12** is reset by either of the methods described, the terminal Self-Test will be executed. The Self-Test will verify the integrity of the display memory, the program memory, non-volatile memory and the associated internal control logic. Upon completion of Self-Test, the terminal will display the "OK" message.

3.13.4 Status Information

The following is a list of status information that appears on the 25th status line, and their locations.

3.14 ANSWERBACK MEMORY PROGRAM

The user selectable Answerback message is programmed into the ADM 12 by first entering the Supervisor Set-Up Mode as described in Section 3.2, page 3-1, and then entering the control sequence <u>CTRL</u>[A]. Any message up to 32 characters may be entered from the keyboard as the answerback message. In order to save the answerback message in non-volatile memory, enter the control sequence <u>CTRL</u>[S] before exiting the Set-Up Mode.

DESCRIPTION .	COLUMN LOCATION	DISPLAY
PAGE DESIGNATOR	1-2	P n n = 1 or 2
CURSOR ROW ADDRESS	4-6	R nn nn = 01 to 48
CURSOR COLUMN ADDRESS	8-11	C nnn nnn= 001-158
PAGE CONFIGURATION	13-16	LONG 1 page 48 x 80
		WIDE 1 page 24 x 158
PROTECT MODE	18-21	PROT
WRITE PROTECT	23-25	WPT
BUSINESS GRAPHICS	27-30	
INSERT MODE	32-35	INSM
PAGE EDIT MODE	37-42	PGEDIT
LINE MODE	44-46	LOC LOCAL MODE (OFFLINE)
		FDX ONLINE - FULL DUPLEX
		HDX ONLINE - HALF DUPLEX
TERMINAL MODE	48-50	CNV CONVERSATION MODE
		BLK BLOCK MODE
DATA CARRIER DETECT	52-54	DCD
CLEAR TO SEND	56-58	CTS
PARITY ERROR	60-61	PE
KEYBOARD LOCKED	63-66	KBLK
AUXILIARY PORT ENABLED	68-71	AUX
		AUXT TRANSPARENT MODE
PROGRAM MODE	73-75	PRG
	77-79	HLD HOLD SCREEN

LEAR SIEGLER, INC. DATA PRODUCTS DIVISION



Figure 3-7. Set-Up Mode Flow Chart

FUNCTION	STANDARD	BUSINESS GRAPHICS
Reset to Normal Video	ESC G 0	ESC G 0
Business Graphics Mode Enable		ESC G 8
Blanking (Set-Up 06, Bit 2 = 0) or Underlining (Set-Up 06, Bit 2 = 1) of Area	ESC G 1	ESC G 9
Blinking of Area	ESC G 2	ESC G A
Blanking (Set-Up 06, Bit 2 = 0) or Underlining (Set-Up 06, Bit 2 = 1) and Blinking of Area	ESC G 3	ESC G B
Reversing of Area	ESC G 4	ESC G C
Blanking (Set-Up 06, Bit 2 = 0) or Underlining (Set-Up 06, Bit 2 = 1) and Reversing of Area	ESC G 5	ESC G D
Blinking and Reversing of Area	ESC G 6	ESC G E
Blanking (Set-Up 06, Bit 2 = 0) or Underlining (Set-Up 06, Bit 2 = 1) and Reversing and Blinking of Area	ESC G 7	ESC G F

Table 3-4. Embedded Attributes Operation Codes

LEAR SIEGLER, INC.

APPENDIX A OPERATORS QUICK REFERENCE CHARTS

DESCRIPTION	DISPLAYED AS	VALUES	DISPLAYED AS		
Keyclick	CLK	No Click CLICK *	N Y		
Status Line Attributes	STS	Normal Blank Blink REVERSE * Reduced Intensity Reduced-Reverse	NRM BLANK BLINK REV RED RED-REV		
Cursor Blink	CRSR BLINK	NON-BLINKING * Blink	N Y		
Cursor Shape	CRSR	BLOCK * Underline	BLK UND		
Sreen Saver	SCRN SAVE	Disabled ENABLED *	N Y		
Smooth Scroll	SMOOTH SCRL	DISABLED * Enabled	N Y		
Horizontal Scroll	HORZ SCRL	By Line BY PAGE *	LN PG		
On Line/Local	ONLN LOCL	ON LINE * Local	ONLN LOCL		
Page Configuration	PG CNFG	TWO IND. PAGES * One Long Page One Wide Page	2PG LNG WDE		
Wraparound at Right Margin	WRAP	YES * No	Y N		
Return Key Configuration	NEWLN	CR * CRLF	N Y		
Block/Conversation	CNV BLK	CONVERSATION MODE *	CNV		
Full/Half Duplex	HDX FDX	Block Mode HALF DUPLEX * Full Duplex	BLK HDX FDX		
Handshake Protocol	HNDSHK	None DTR X-ON/X-OFF *	NO DTR XON		

Table A-1. Set-Up Mode Functions (To enter Set-Up Mode, see Section 3.2, page 3-1.)

* Indicates factory default condition,

LEAR SIEGLER, INC.

DESCRIPTION	DISPLAYED AS	VALUES	DISPLAYED AS
Baud Rate	BPS	300 Baud 600 Baud 1200 Baud 2400 Baud 4800 Baud 9600 BAUD * 1 9 200 Baud	300 600 1200 2400 4800 9600 19200
Parity Select	ΡΤΥ	No Parity PARITY *	N Y
Parity Type	ΡΤΥ	EVEN * Odd	EV OD
Bit 8 = 1 or 0	BIT 8	0 1 *	0 1

Table A-1. Set-Up Mode Functions (continued)

* Indicates factory default condition.

Table A-2. Binary Set-Up Functions

The Set-Up features can be down-line loaded by the host computer using the following command sequence:

ESC { AA HH

Where "AA" is the Hex encoded Set-Up byte to be addressed and HH is the Hex encoded desired set-up condition.

OR

Set-Up functions can be configured from the keyboard by entering Supervisor Set-Up SHIFT CTRL SETUP then entering CTRL B

SET-UP BYTE	BITS 7 6 5 4 3 2 1 0	DESCRIPTION	VALUES
00H 00 (Hex)		0 Full Screen Reverse	1 = Reverse 0 = NORMAL*
		1 Load Cursor Skip	1 = SKIP* 0 = Normal
		2 Print Disable	1 = Disable 0 = ENABLE *
		3 Keyclick	1 = CLICK * 0 = No Click
		4 Wraparound at Right Margin	1 = No 0 = YES*
		5 New Line With CR	1 = Yes 0 = NO*
		6 Line Erase With CR	1 = Yes 0 = NO*
		7 Inhibit Supervisor Set-Up**	1 = Yes 0 = NO.*

NOTE:

**CANNOT BE ENTERED FROM THE KEYBOARD ONCE THIS MODE IS ENABLED. DEFAULT VALUES DISPLAYED IN UPPER CASE.

01H 01 (Hex) 0 Status Line Attributes 000 = Normal 001 = Blank 001 = Blank 001 = Blank 011 = REVERSE* 100 = Reduced Intensity 101 = Reduced Intensity 100 = Reduced Intensity 100 = Reduced Intensity 101 = Reduced Intensity 100 = Reduced Intensity 100 = Reduced Intensity 101 = Reduced Intensity 101 = Reduced Intensity 100 = Roduced Intens	SET-UP BYTE	BITS 7 6 5 4 3 2 1 0	DESCRIPTION	VALUES
From Host 0 = N0* 4 Home Cursor With Protect 1 = YES* 0 = No 1 = CR LF 0 = CR* 0 = CR* 6 Page Number Designation 1 = 0/1 * 0 = 1/2 1 = Yes 7 New Line With LF In 1 = Yes 0 = NO* 0 = NO 0 = NO 1 = 0/1 * 0 = NO* 0 = 1/2 7 New Line With LF In 1 = Yes 0 = NO* 0 = NO* 0 = NO* 1 = Yes 0 = NO* 0 = NO* 0 = NO* 0 = NO* 0 = VI 0 = NO* 0 = NO* 0 = NO* 0 = NO* 0 = NO* 0 = NO* 1 = Yes 0 = NO* 2 Clear Command Control 1 = ENABLES 4 CLEAR COMMANDS* 0 = Enables 2 Clear Commands 1 = Disable 0 = ENABLE* 0 = ENABLE* 1 = Disable			1	001 = Blank 010 = Blink 011 = REVERSE * 100 = Reduced Intensity 101 = Reduced-Reverse 110 = Unallowed
0 = No 5 Return Key Configuration 6 Page Number Designation 1 = 0/1 * 0 = 1/2 7 New Line With LF In 1 = Yes 0 = NO* 0 = No 1 = 0/1 * 0 = NO* 0 = NO* 1 = Yes 0 = NO* 0 Erase Line Resets Write 1 = Yes 0 = NO* 0 Erase Line Resets Write 1 = Yes 0 = NO* 1 Erase Page Resets Write 1 = Yes 0 = NO* 2 Clear Command Control 3 Scrolling Disable 1 = Disable 0 = ENABLE*				
0 = CR * 6 Page Number Designation 1 = 0/1 * 0 = 1/2 7 New Line With LF In Protect Mode 1 = Yes 0 = NO* 02H 0 Erase Line Resets Write Protect 1 = Yes 02(Hex) 0 Erase Page Resets Write Protect 1 = Yes 0 Erase Page Resets Write Protect 1 = Yes 0 Erase Page Resets Write Protect 1 = Yes 0 = NO * 1 = Stable S 4 CLEAR COMMANDS * 0 = Enables 2 Clear Commands 1 = Disable 0 = Enables 2 Clear Commands 1 = Disable 0 = Enables 2 Clear Commands 1 = Disable			4 Home Cursor With Protect	
0 1/2 7 New Line With LF In Protect Mode 1 = Yes 0 = NO* 7 6 5 4 3 2 1 02H 02 (Hex) 0 Erase Line Resets Write Protect 1 = Yes 0 = NO* 0 = NO* 1 Erase Page Resets Write Protect 1 = Yes 0 = NO* 0 = NO* 2 Clear Command Control 3 1 = ENABLES 4 CLEAR COMMANDS* 0 = Enables 2 Clear Commands 3 Scrolling Disable 1 = Disable 0 = ENABLE*			5 Return Key Configuration	
Protect Mode D = NO* 7 6 5 4 3 2 1 0 0 Erase Line Resets Write Protect 1 = Yes 0 = NO* 02H 02 (Hex) 0 Erase Line Resets Write Protect 1 = Yes 0 = NO* 1 Erase Page Resets Write Protect 1 = Yes 0 = NO* 2 Clear Command Control 1 = ENABLES 4 CLEAR COMMANDS* 0 = Enables 2 Clear Commands 3 Scrolling Disable 1 = Disable 0 = ENABLE*			6 Page Number Designation	
02H 0 Erase Line Resets Write 1 = Yes 02 (Hex) 1 Erase Page Resets Write 1 = Yes 1 Erase Page Resets Write 1 = Yes 0 NO* 2 2 Clear Command Control 1 = ENABLES 4 CLEAR COMMANDS * 0 = Enables 2 Clear Commands 3 Scrolling Disable 1 = Disable 0 = ENABLE * 0 = ENABLE *				
02 (Hex) Protect 0 = NO* 1 Erase Page Resets Write Protect 1 = Yes 0 = NO* 2 Clear Command Control 1 = ENABLES 4 CLEAR COMMANDS* 3 Scrolling Disable 1 = Disable 0 = ENABLE*		7 6 5 4 3 2 1 0		
Protect 0 = NO* 2 Clear Command Control 1 = ENABLES 4 CLEAR COMMANDS* 0 = Enables 2 Clear Commands 1 = Disable 0 = NO*				
3 Scrolling Disable COMMANDS * 0 = Enables 2 Clear Commands 1 = Disable 0 = ENABLE *				
3 Scrolling Disable 1 = Disable 0 = ENABLE*			2 Clear Command Control	COMMANDS *
4 Inhibit New Line Fills 1 = Yes			3 Scrolling Disable	1 = Disable
With Send 0 = NO*			4 Inhibit New Line Fills With Send	1 = Yes 0 = NO*
5 Protect Field Delimiter Placement 1 = Embedded After Protected Field 0 = EMBEDDED BEFORE*				Field 0 = EMBEDDED BEFORE*
PROTECTED FIELD				PROTECTED FIELD
6 Reserved			6 Reserved	
NOTE:			7 Reserved	

Table A-2. Binary Set-Up Functions (continued)

NOTE:

DEFAULT VALUES DISPLAYED IN UPPER CASE AND BY AN *.

LEAR SIEGLER, INC. DATA PRODUCTS DIVISION

SET-UP BYTE	BITS 7 6 5 4 3 2 1 0	DESCRIPTION	VALUES		
03H 03 (Hex)		0 ETX Terminates Write Protect	1 = Yes 0 = NO*		
		1 Cursor Shape	1 = Underline 0 = BLOCK*		
		2 Cursor Blink	1 = Blink 0 = NON-BLINKING*		
		3 Print Attribute	1 = SPACE PRINTED * 0 = Attribute Printed		
		4 New Line After Send In Protect Mode	1 = Yes 0 = NO *		
		5 On Line/Local Extention (Aux) Busy Level	1 = LOW * 0 = High		
		6 Block/Conversation	1 = Block Mode 0 = CONVERSATION MODE*		
		7 Screen Saver	1 = ENABLED* 0 = Disabled		
	7 6 5 4 3 2 1 0				
04H 04 (Hex)	}_{{	0 Baud Rate 1 2	000 = 300 Baud 001 = 600 Baud 010 = 1200 Baud 011 = 2400 Baud 100 = 4800 Baud 101 = 9600 BAUD * 110 = 19200 Baud 111 = Unallowed		
		3 Parity Enabled	1 = YES * 0 = No		
		4 Parity Sense If Enabled	1 = Odd 0 = EVEN*		
		5 Bit 8 If No Parity	1 = HIGH '1' * 0 = Low '0'		
		6 Duplex Mode	1 = Full Duplex 0 = HALF DUPLEX *		
NOTE:		7 On Line/Local	1 = Local 0 = ON LINE *		

Table A-2. Binary Set-Up Functions (continued)

NOTE: DEFAULT VALUES DISPLAYED IN UPPER CASE.

SET-UP	BITS		
BYTE	7 6 5 4 3 2 1 0	DESCRIPTION	VALUES
05H 05 (Hex)		0 Page Display Configuration 1	00 = 2 INDEPENDENT* PAGES 24 x 80 01 = 1 Page 48 x 80 (Double Long) 10 = 1 Page 24 x 158 (Double Wide) 11 = Unallowed
	{	2 Reserved 3	
	{	4 Handshake Mode 5	00 = No Handshaking 01 = DTR Handshaking 10 = XON/XOFF HANDSHAKING* 11 - Unallowed
		6 Vertical Refresh Rate	1 = 50 Hz 0 = 60 HZ*
		7 Embedded Mode	1 = Embedded Attributes 0 = NON-EMBEDDED*
	7 6 5 4 3 2 1 0		
06H 06 (Hex)		0 Horizontal Scroll	1 = PAGE * 0 = Line
		1 Smooth Scroll	1 = Smooth 0 = JUMP*
		2 ESC G1 Attribute In Embedded Mode	1 = UNDERLINE * 0 = Blank
		3 Non-Programmable Edit Keys	1 = LOCAL* 0 = Duplex
		4 Reserved 5 6 7	

Table A-2. Binary Set-Up Functions (continued)

NOTE:

DEFAULT VALUES DISPLAYED IN UPPER CASE.

LEAR SIEGLER, INC.

HEX		2		3		4	5			6		7
0			0	CLEAR TABS ALL	@	PRINT PROTECTED SPACES	Ρ	FORMATTED PRINT	`		р	UNFORMAT Print
1	!	PROGRAM Function Keys*	1	SET TAB	A	AUXILIARY Control*	Q	INSERT CHARACTER	а		q	SET INSERT MODE
2	,,	KEYBOARD ENABLE	2	CLEAR TAB	в	SET BLOCK MODE	R	DELETE LINE	b		r	CLEAR INSERT MODE
3	Ħ	KEYBOARD DISABLE	3		с	SET Conversation Mode	S	SEND Message Foreground	с	SET SCROLL REGION*	s	SEND MESSAGE ALL
4	\$		4	SEND Line Foreground	D	SET DUPLEX*	т	ERASE LINE FILL 1	d		t	ERASE LINE FILL 2
5	%		5	SEND PAGE FOREGROUND	E	INSERT LINE	υ	SET PROGRAM MODE	e	SET Smooth Scroll	u	CLEAR Program Mode
6	&	SET PROTECT MODE	6	SEND LINE ALL	F	WRITE MESSAGE LINE #1*	v		f	SET Jump Scroll	v	SET LONG PAGE
7	,	CLEAR PROTECT MODE	7	SEND PAGE ALL	G	ATTRIBUTE Control*	w	DELETE CHARACTER	g	DISPLAY Host Message #1	w	CLEAR LONG PAGE
8	(CLEAR WRITE PROTECT	8	WRITE SOM	н	ALIGNMENT PATTERN	х	CLEAR Program Mode	h	DISPLAY Host Message #2	x	
9)	SET WRITE Protect	9	WRITE EOM	1	BACK Tab	Y	ERASE PAGE FILL 1	i	HORIZONTAL TAB	у	ERASE PAGE FILL 2
A	÷	CLEAR All to Fill 2	:	CLEAR Foreground To Fill 2	∕ J	PAGE BACK	z	CLEAR ALL MODES	j		z	PRINT Line Formatted
в	+	CLEAR All to Fill 1	:	CLEAR Foreground To Fill 1	к	PAGE Forward	[k		{	WRITE SET UP BYTE*
С	•	CLEAR ALL TO PROTECT SPACES	<		Ļ	SET GRAPHICS MODE	1	ENVOKE Function Key*	1	WRITE MESSAGE LINE 2		SELECT Baud Rate*
D	-	ADDRESS CURSOR* P, R, C	11	ADDRESS Cursor* R, C	м]		m	WRITE Multiple Characters*	}	LINK Function Keys*
E		CHANGE Personality*	$^{>}$		N	SET PAGE EDIT	Λ	SET Line 25 Mode*	n		2	CURSOR Video Control*
F	/	READ CURSOR P, R, C	?	READ Cursor R, C	0	CLEAR Page Edit			ο	RESET/ DEFAULT/ VERSION*	DEL	

Table A-3. ADM 12 Escape Sequences

*MULTIPLE CHARACTERS REQUIRED.
OR:					SCII 4	COL ASCII] _ 0 1 :f D t	• D :	4° 1
	ESC =	PAC ASC		ROW COL ASCII ASCII	Page .		= 0 or 1 if Page # = 1 or 2 if Page #	# Designa # Designa	tion = 1
		POSI	TION		POSI	TION		POSIT	TION
AS	CII CODES	ROW	COL	ASCII CODES	ROW	COL	ASCII CODES	ROW	COL
ESC	C = SPACE	1	1	ESC = ;		28	ESC = V		55
		2	2			29			56
		3	3			30			57
	#	4	4			31			58 50
	\$	5	5			32			59 60
	8	6 7	6 7	@ A		33 34			61
	&	8	8	B		35			62
		9	9			36			63
		10	10			37			64
	*	11	11	E		38			65
		12	12	F		39	a b		66
	,	13	13	G		40	Ь		67
	-	14	14	Н		41	C		68
		15	15			42			69
		16	16			43	e		70
	0	17	17	K		44	E F		71
	1	18	18			45			72
	2	19	19	M		46	h i		73
	3	20	20			47			74 75
		21	21	0 P		48			76
		22 23	22 23			49	J J		77
	5 6 7	23	23			50 51			78
	8	67	25			51			79
	8		26			53			80
ES			27			54			

Table A-4. Reference Chart of Absolute Cursor Positions

A-8

Table A-4. Reference Chart of Absolute Cursor Positions (continued)

				$SC = \begin{bmatrix} ROV \\ ASC \end{bmatrix}$ (for Wide P	II A	COL SCII play)				
	POSI	TION			POSI	TION			POSI	LION
ASCII CODES	ROW	COL	AS	CII CODES	ROW	COL	AS	CII CODES	ROW	COL
ESC = SPACE		81	ESC			108	ESC	I = V W	7	135
		82 83		-		109 110			8 9	136 137
		84				110		X Y	10	138
\$		85		?		112		Z	10	139
		86		0		113			12	140
× &		87		A		114			13	141
		88				115			14	142
		89		B		116			15	143
		90		D		117			16	144
		91		Ε		118			17	145
		92				119		a	18	146
		93		G		120		Ь	19	147
		94		H		121		C	20	148
		95		I		122		٩	21	149
		96		X J		123		e	22	150
		97				124		F	23	151
		98		L		125		g	24	152 153
2		99				126		i		155
3		100 101				127				155
		101		0 P	1	128				155
4 5 6		102		Q	2	129 130		j k 1		157
		104			3	131		Ē		158
8		105		S	4	132		n		
7 8 9		106		RS	<u>ີ</u> 5	133	ESC			
ESC = :		107	ESC		6	134				

	T																		
B7 B6	0	0	0	1		0	1		1	0		1	0		1	1		1	1
BITS B5	0	1		•	0			1			0			1			0		1
D4 D2 D2 D1	1	TROL				BER	-			UPP	ER	CA	ASE			LO۱	NE	R C	ASE
B4 B3 B2 B1	0		40			60			100			120			140			160	
0 0 0 0	NUL	DLE 10 16	1	SP		30	0	48	40	@	64	50	Ρ	80	60	١	96	70	P 112
	1	21	41			61			101	_		121	~		141			161	
0 0 0 1	,SOH,	DC1	21	!	33	31	1	49	41	Α	65	51	Q	81	61	а	97	71	q 113
		22	42			62	0		102	в		122	D		142	h		162	
0 0 1 0	STX 2	DC2 12 18	22		34	32	2	50	42			52		82	62				r 114
0011	втх	²³ DC3	43	#		63	3		103	С		123	S		143	С		163	S
		.13 19 24	23 44		35	33 64		51	43 104		67	53 124		83	63 144		99	73 164	115
0 1 0 0	EOT	DC4		\$			4			D			т			d			t
	4 4 5	14 20 25	24 45		36	34 65		52	44 105		68	54 125		84	64 145		100	74 165	1 16
0 1 0 1	ENQ			%			5			Ε			U			е			u
	5 5 6	15 21 26	25 46		37	35 66		53	45 106		69	55 126		85	65 146		101	75 166	117
0 1 1 0	ACK	SYN		&	~~	200	6		40	F	70		V			f	102	76	V 118
······	6 6 7	16 22 27	47	,	38	36 67		54	46 107		/0	56 127		80	66 147		102	167	110
0 1 1 1	BEL	ETB	27	'	39	37	7	55	47	G	71		W	87	67	g	103	77	W 119
_	10 🔶	30	50	,		70	_		110			130			150			170	
1000	BS 8	CAN 18 24	28	(40	38	8	56	48	Η	72	58	Χ	83	68	h	104	78	X 120
1001	11 ТАВ НТ	³¹ EM	51)		71	9		111	1		131	Y		151	i		171	v
	9 9	19 25			41	39		57	49		73		-	89		-	105	79 172	У 121
1010	¹² LF ⁺	³² SUB	52	*		72	:		112	J		132	z		152	;		172	z
	A 10		2A 53		42	3A 73		58	4A 113		74	5A 133		90	6A 153	J	106	7A 173	122
1011	∣҄ѵт҅	ESC		+			;			Κ			Γ			k			{
	B 11 14 →	1B 27 34	28 54		43	3B 74		59	4B 114		75	5B 134		91	6B 154		107	7B 174	۱ ₂₃
1 1 0 _c 0	FF	FS		,			<	60		L	70	50	١			I	100	70	
	15 RTN		2C 55		44	3C 75		60	4C 115		76	5C 135		92	6C 155		108	7C 175	124
1 1 0 1		1D 29	2D		45	3D	==	61	4D	Μ	77	5D]	93	6D	m	109		}
	16	36 HOME				76	_		116	N 1		136			156			176	
1 1 1 0	SO E 14	1E 30	2E	•	46	3E	>		4E	N		5E	Λ	94	6E	n	110	7E	\sim_{126}
1 1 1 1	¹⁷ SI	37 NEW LINE US	57	1		77	2	JNL	117	0		137			157	0			DEL
1 1 1 1		1F 31	2F	/	47	ЗF	1	63	4F	<u> </u>	79	5F		95	6F	0	111	RU 7F	BOUT 127
				oc	tal [1	0		+	LSI	CUR	SOR	CONT	ROL	coc	ÞΕ				
		KEY					3S			CII ch									
				h	ex 8	3		8	dec	imal									

Table A-5. ASCII Control Code Chart

Table A-6. Displayable USASCII Character Set and Control Codes

			CONTROL	- CHARACTERS		DISPL	AYABLE	CHARA	CTERS	
HEX		1ST	0	1	2	3	4	5	6	7
BYTE 2ND	bits ⁴ 321	BITS 7 ₆₅	0 ₀₀	0 ₀₁	0 ₁₀	0 ₁₁	1 ₀₀	¹ 01	1 ₁₀	1 ₁₁
0	0000		NUL	DLE		0	@	Р	,	р
1	0001		SOH	DC1	!	1	А	٥	а	q
2	0010		STX	DC2	"	2	В	R	b	r
3	0011		ETX	DC3	#	3	с	S ·	с	s
4	0100		EOT	DC4	\$	4	D	т	d	t
5	0101		ENQ	ΝΑΚ	%	5	E	U	е	u
6	0110		АСК	SYN	&	6	F	V	f	v
7	0111		BEEP	ЕТВ	,	7	G	w	g	w
8	1000		BS (←)	CAN	(8	н	×	h	×
9	1001		(SKIP) HT	EM)	9	I	Y	i	У
A	1010		LF (↓)	SUB	*	:	J	z	j	ż
В	1011		VT (†)	ESC	+	;	к	[k	{
С	1100		FF (→)	FS	,	<	L	\	1	1
D	1101		CR	GS	-		м]	m	}
E	1110		SO	(HOME) RS	•	>	N	^	n	\sim
F	1111		SI	(NEW LINE) US	/	?	0		0	DEL

USE CTRL KEY WITH DISPLAYABLE CHARACTER KEYS TO PRODUCE CONTROL CODES

00 NUL	91 SOH	02 STX	03 ETX	04 EOT 	05 ENQ	06 ACK	07 BEL	08 BS []] . .	09 HT	0A LF 	0B VT :::	9CFF 1.1.	0D CR	0E SO 	0F SI
10 DLE	11 DC1	12 DC2	13 DC3	14 DC4	15 NAK	16 SYN	17 ETB	18 CAN	19 EM	1A SUB I.I.	1B ESC 	105 I	10% I.:. .I'I	1E RS !::: ^{1.1*1}	1F US
20	21 : :	22 ::	23 	24 ::: 	25 	26	27 	28 	29	2A	2B 	2C 	2D	2E :	2F
30	31	32 	33 	34	35	36 	37 	38	39	3A :: ::	3B :: 	3C	3D	3E	3F :: :
40	41	42	43 	44	45	46	47 	48	49	4A	4B	4C	4D	4E	4F
50	51	52	53 :	54	55	56 : : : : : :	57	58 : .: : .:	59 ::	5A	5B	5C 	5D	5E .∵.	5F
60 ".	61	62 	63 ::	64	65	66	67	68 	69	6A 	6B	6C	6D	6E	6F
70	71	72 	73	74	75	76 :: 	77	78 	79	7A	7B	7C : :	7D	7E .∵	7F

Table A-7. 128 Character ASCII Format, with Hex Codes



Table A-8. Business Graphics Character Set

Table A-9. International Keyboard Layouts

ADM 12 SWEDISH/FINNISH QWERTY KEYBOARD LAYOUT

SET UP STATUS	F 1	⁷ F	18 2	19 3	F 420	F	21 5 F	22 6 F	23 7	F 8 8	F 9 9	1 1	26 F	27 11	F 28 12	F 29 F 13	F 30 F 14	F ³¹ 15	F 16	LINE Insert Char	PAGE ERASE LINE	LINE Delete Char	pg edt Insert	BREAK
LOCAL	1		2	# 3	\$	%	p l	& 6	/	()	=	Т	?	•	>	BACK SPACE	SEND	SEND MSG	PRINT	ТАВ	CLEA DELET	R AUX Te port	SCROLL
ESCAPE	<u> </u>	Q	w	E	Ť	R	T	Y	U			0	P	Á				PAGE NEW Line	GRAPH Norm	REV BK GRD Norm	7	8	9	-
CTRL	CAP Lock	A	S		D	F	G	н	J	J	к	L	Ö	1	Ă	RETURN			1		4	5	6	,
	SHIF	т	Z	X	C		v	в	Ν	м	;		:	Ŧ	s	HIFT	* @	-	HOME	→	1	2	3	E N T
•																			LINE FEED ↓			0		Ë R

ADM 12 NORWEGIAN-QWERTY KEYBOARD LAYOUT

SET UP STATUS		17 1	F _ 18 _ 2	1 F	19 3	F 4 20	F 2	1 1	22 6 F	23 7	24 8	F 9 9	1 1	26 10 F	27 11	F 28 12	1 1-	F 30 14	F 31 F 15	F ³² 16	LINE INSERT CHAR	PAGE ERASE LINE	LINE DELETE Char	G EDT NSERT	BREAK
LOCAL ESCAPE ESCAPE			" 2	#		\$ 4	% 5		84 6	, 7	(8) 9			?	~	=	BACK SPACE	SEND	SEND MSG	PRINT	TAB	CLEAR DELETE	AUX Port	SCROLL
TA		Q		w	E	F	2	T	Y	U	1		0	P	Å	1	+;		PAGE NEW LINE	GRAPH Norm	BK GRD Norm	7	8	9	-
CTRL	CAP Lock		A	S		>	F	G	Н	J		ĸ	L	Ø	1	Æ	RETURI			+		4	5	6	,
	SHI	IFT	Z	z	X	с		v	в	N	М	<		>	*		SHIFT	、 @	-	номе	→	1	2	3	E N T
-																				LINE FEED			0		E R

ADM 12 SPANISH-QWERTY KEYBOARD LAYOUT

SET UP STATUS		7 F	18 2	F 1 3	⁹ F	20 4	F 521	F 622	F	23 7 F	24 8	25 9	F 26 10	F 1	27 1 F	28 12	F 29 13	F 30 F 14	F 31 15	F ³² 16	LINE INSERT Char	PAGE ERASE LINE	LINE Delete Char	g edt Nsert	BREAK
LOCAL ESCAPE ESCAPE	f 1		,, 2	\$ 3		% 4	& 5	(6]		8	ý 9	= 0	?		*	> <	BACK SPACE	SEND	SEND MSG	PRINT REV	TAB	CLEAR DELETE	AUX Port	SCROLL
TA	<u> </u>	Q	W	'	E	R	1		Y	U		0		P	! ! i				PAGE NEW LINE	GRAPH Norm	BK GRD Norm	7	8	9	-
CTRL	CAP Lock	A		S	D		F	G	Η	J	ĸ		L	Ñ	ç		RETURN			†		4	5	6	,
	SHIF	т	Z		x	С	v	B		N	м	;	:		• -	SH	IIFT	 ,	•	HOME	->	1	2	3	E N
•							•						•						L	LINE FEED↓			0		Ē

Table A-9. International Keyboard Layouts (continued)

F 27 F 29 F 26 F 30 F 22 F 23 F 24 LINE INSERT CHAR LINE DELETE Char 25 9 F 32 16 PAGE ERASE F 17 1 F 3 SET UP STATUS 18 20 28 31 PG EDT INSERT F F F F F BREAK 11 2 4 6 7 8 10 12 13 14 15 LINE SEND MSG CLEAR DELETE AUX Port SEND PRINT SCROLL TAB LOCAL ESCAPE ESCAPE " 2 § 3 < > BACK Space % 5 & 6 (8 **?** β /7 1 \$ 4) 9 = 0 ۱ i , PAGE REV GRAPH Norm NEW BK GRD NORM * 7 8 9 Ε Т Ζ 0 Ρ Ü TAB Q W R U 1 + CAP Lock ö S Α Ä CTRL D F G Η J Κ L RETURN ŧ 4 5 6 , \wedge ; : SHIFT γ X С A, -SHIFT HOME -+ 1 В Ν Μ 4-2 3 E N T # , Ė LINE FEED ↓ 0 .

ADM 12 GERMAN-QWERTZ KEYBOARD LAYOUT

ADM 12 FRENCH-AZERTY KEYBOARD LAYOUT

SET UP STATUS	F 1	⁷ F	18 2	F 3	Ð F	20 4	F 21 5	F	22 5 F	23 7	F 824	F g		F 26 10	F	27 11 F	28 12	F 29 F 13	F 30 F 14	F 31 15	F	32 16	LINE INSERT CHAR	PAG ERAS Line	ED	LINE Elete Char	PG EDT Insert	BREAK
LOCAL ESCAPE ESCAPE	1		2 é	3	Τ	4	5		5	7 è	8	9 ç	Τ	0 à	°			\$ #	BACK SPACE	SE	-	SEND MSG	PRINT	Г	AB	CLEAR DELET	AUX Port	SCROLL
TA		A	Z	:	E	R	Ì	T	Y	U			0		P	 ^		、		PAI NE LIN	W IE	GRAPH NORM	BK GRD NORM		7	8	9	-
CTRL	CAP Lock	Q		S	D		F	G	Н		J	Κ	l	-	М	% ù		RETURN				t			4	5	6	,
	SHIF	т	W		x	С	V	'	в	N	?		•	:	,	+=	S	HIFT	><	•	-	HOME	->		1	2	3	E N T
·																						LINE FEED ↓			()		R

ADM 12 DANISH-QWERTY KEYBOARD LAYOUT

SET UP STATUS		17 F	- 18 2	F	19 3	F 4) F	21 5	F 6	2 F	23 7	F 824	F	25	26 10	F 1	1 F	28 12	29 13	F 30 14	F ³¹ 15	F ³² 16	LINE INSERT CHAR	PAGE ERASE LINE	LINE Delete Char	PG EDT Insert	BREAK
LOCAL ESCAPE ESCAPE	!	Τ	" 2			\$ 4	Τ	% 5	& 6	Τ	ż	(8) 9		0	=	Â		*	BACK SPACE	SEND	SEND MSG	PRINT	ТАВ	CLEA DELE	AR AUX Te port	SCROLL
TA		Q		w	Ε		R	T	•	Y	U		1	0		>	Å	 + ;			PAGE NEW LINE	GRAPH NORM	BK GRD	7	8	9	-
CTRL	CAP Lock	A		S	1	D	F	:	G	н		J	K	L		Æ	Ø	RE	TURN		.	+		4	5	6	,
	SHI	FT	Z	z	X		2	v		3	N	м		< ,	>		?	SHIF	т	` @	-	HOME	-+	1	2	3	ENT
L																						LINE FEED			0		Ē

Table A-10. Model Configuration Identification

The configuration number is used to identify the exact configuration of your terminal. It appears on the back of all units and also on the outside of the shipping box. A key to the individual configuration digits is listed below:

	T DESIGNATI		$- \frac{x_5}{1} \frac{x_6 x_7}{1}$	× ₈	$\frac{x_9 x_{10}}{10} \frac{x_1}{10}$	1 X ₁₂
UNIT CO	NFIGURATIO					
(VOL	TAGE)					
LOGIC B	OARD CONFIG	GURATION				
TILT/SW	IVEL					
КЕҮВОА	RD					
<u>×₅</u>	CODE	<u>136887</u> <u>U</u>	NIT CONFIGURATION	<u> </u>	DASH NO.	
1 2	115V 230V	12'' MONITOR (GREEN, P3 12'' MONITOR (GREEN, P3			-001 -002	
3	115V 230V	12" MONITOR (AMBER) 12" MONITOR (AMBER)			-003 -004	
5 6	115V 230V	14" MONITOR (GREEN, P3 14" MONITOR (GREEN, P3			-005 -006	
7	230V 115V	14" MONITOR (AMBER))))		-007	
8 9	230V 115V	14" MONITOR (AMBER) 12" MONITOR (GREEN, P3	21) CSA		-008 -009	
A	115V 115V	12" MONITOR (GREEN, PC	CSA		-009 -00A	
			i			
$\frac{x_6}{2}$ $\frac{x_7}{2}$	CODE	136908 LOGIC	BOARD CONFIGURA	TION	DASH NO.	
0 1	STD BD	(BUSINESS GRAPHICS)			-001	
0 2 0 3	STD BD STD BD	(BUSINESS GRAPHICS) (BUSINESS GRAPHICS)	CURRENT LOOP	RS-422	-002 -003	
0 4	AZERTY	(BUSINESS GRAPHICS)			-004	
05	AZERTY AZERTY	(BUSINESS GRAPHICS) (BUSINESS GRAPHICS)	CURRENT LOOP	RS-422	-005 -006	
0 7	QWERTZ	(BUSINESS GRAPHICS)		110-422	-007	
0 8	QWERTZ	(BUSINESS GRAPHICS)	CURRENT LOOP		-008	
0 9	QWERTZ UK	(BUSINESS GRAPHICS) (BUSINESS GRAPHICS)		RS-422	-009 -010	
	UK	(BUSINESS GRAPHICS)	CURRENT LOOP		-010	
1 2	UK	(BUSINESS GRAPHICS)		RS-422	-012	
1 3	SPANISH SPANISH	(BUSINESS GRAPHICS) (BUSINESS GRAPHICS)	CURRENT LOOP		-013 -014	
1 5	SPANISH	(BUSINESS GRAPHICS)		RS-422	-014	
1 6	NORW	(BUSINESS GRAPHICS)			-016	
1 7	NORW NORW	(BUSINESS GRAPHICS) (BUSINESS GRAPHICS)	CURRENT LOOP	RS-422	-017 -018	

.

 LEAR	SIEGLER, INC.	,
DATA	PRODUCTS DIVISIO	N

Table A-10. Model Configuration Identification (continued)

<u>x₆ x₇</u>	CODE	136908 CON	LOGIC BOARD FIGURATION (CONT'D)	DASH NO.
1 9 2 0 2 1 2 2 2 3 2 3 2 4	DANISH DANISH DANISH SWED/FIN SWED/FIN SWED/FIN	(BUSINESS GRAPHICS) (BUSINESS GRAPHICS) (BUSINESS GRAPHICS) (BUSINESS GRAPHICS) (BUSINESS GRAPHICS) (BUSINESS GRAPHICS)	CURRENT LOOP	RS-422 RS-422	-019 -020 -021 -022 -023 -024
1 X ₈	<u>CODE</u> TILT/SWIVE		TILT/SWIVEL		-001
X ₁₁ CODE 136678 KEYBOARD ASSY, LOW PROFILE					
1 2	STD KEYBO TILT/SWIVE				-101 -102
6 7	STD KEYBO TILT/SWIVE	• • • • • • • • • • • •			-006 -007

INDEX

SUBJECT	A
Absolute Cursor Positioning Absolute Cursor Positioning Operation Code	
Answerback	
ASCII Control Code Chart	
	В
Block Mode	
Busy HI/LOW	1-7,3-31,3-33
	C
Clear To Spaces	
Conversation Mode	
Cursor Control	
	D
Data Transmission	
Default Function Keys	
Delete Character Delete Line	
Display Character Format	
	E
Erase To-End-Of-Line	
Escape	
Liscape Sequences	
	F-G
Formatted Print	
Function Keys	
	Н
Half Duplex	
Handshake	

SUBJECT

PAGE

I-J
Insert Character
Insert Line
Installation
Interfaces:
RS-232C
RS-422
International Character Sets
K
Keyboard Enable/Disable 3-19
Keyboard Layout
Keyboard Operation
Keystroke Conventions
L
Line Print
Line Voltage
M-N
Modem Port
Modifier Keys 3-10
New Line
0
On-Line/Off-Line (Local)
Options
P-Q
Page Print 3-12
Parity Error (PE)
Personality Characters
Power-On Reset
Power Turn-On
Print
Programmable Function Keys
Program Mode
R
Read Cursor 3-22 Reset Terminal 3-26,3-36
Reset Terminal
$RS-232C Interface \dots 2-2, 2-4, 2-6$
1.0-2020 111011200

SUBJECT

PAGE

Screen Saver	
Scrolling	
Self-Test	
Send Line	
Send Message	
Send Page	
Set-Up Mode	1-2,2-1,3-1,3-2,3-3,3-4
Specifications	1-4
Status Key	
Status Line	
Site Requirements	
Tabs	
Transparent Print	

U-V

Unformatted Print	3-12,3-24
Upline	3-8,3-10,3-17
Visual Attributes 1	-4,1-5,3-25,3-33

W-X-Y-Z

Word Structure	1-7
Wraparound	
Write Normal Intensity	3-33
Write Reduced Intensity	3-33



PUBLICATION TITLE PUBLICATION NO. PUBLICATION NO. YOUR NAME YOUR NAME TITLE COMPANY ADDRESS	PUBLICATION NO. DATE YOUR NAME
SUGGESTIONS	
	na shana ka

